## SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

## **SCHEDULE 13D**

(Rule 13d-101)

# INFORMATION TO BE INCLUDED IN STATEMENTS FILED PURSUANT TO RULE 13d-1(a) AND AMENDMENTS THERETO FILED PURSUANT TO RULE 13d-2(a)

(Amendment No. \_\_\_)\*

Artesyn Technologies, Inc.		
(Name of Issuer)		
Common Stock, Par Value \$0.01 Per Share		
(Title of Class of Securities)		
043127109		
(CUSIP Number)		
Harley M. Smith, Esq.		
8000 W. Florissant Avenue		
P.O. Box 4100		
St. Louis, MO 63136-8506		
(314) 553-3800		
(Name, Address and Telephone Number of		
Person Authorized to Receive Notices		
and Communications)		
Copies to:		
Christopher Mayer, Esq.		
Davis Polk & Wardwell		
450 Lexington Avenue		
New York, New York 10017		
(212) 450-4000		
February 1, 2006		
(Date of Event which Requires Filing of this Statement)		

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose Section 18 of the Securities Exchange Act of 1934 or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of Rule 13d-1(e), 13d-1(f) or 13d-1(g), check the following box. o

Note. Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See Rule 13d-7 for other parties to whom copies are to be sent.

CUSIP No. <u>043127109</u> 13D Page 2 of 8 Pages

1 NAME OF REPORTING PERSONS I.R.S. IDENTIFICATION NOS. OF ABOVE PERSONS (ENTITIES ONLY)

<sup>\*</sup> The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

	EMERSON ELECTRIC CO.			
	43-0259330			
2	(a) o			(a) o (b) x
3	SEC USE ONLY			
4	4 SOURCE OF FUNDS			
	NOT APPLICABLE			
5	CHECK BOX IF DISCLOSURE OF LEGAL PROCEEDINGS IS REQUIRED PURSUANT TO ITEM 2(d) or 2(e) o			
6	CITIZENSHIP OR PLACE OF ORGANIZATION			
	MISSOURI			
			SOLE VOTING POWER	
			7,939,749 – See Item 5	
	NUMBER OF SHARES BENEFICIALLY OWNED BY	8	SHARED VOTING POWER	
	EACH REPORTING PERSON WITH	9	SOLE DISPOSITIVE POWER	
			SHARED DISPOSITIVE POWER	
11	AGGREGATE AMOUNT BENEFICIALLY O	WNE	D BY EACH REPORTING PERSON	
	7,939,749 – See Item 5			
12	CHECK BOX IF THE AGGREGATE AMOUNT IN ROW (11) EXCLUDES CERTAIN SHARES  o			
13	PERCENT OF CLASS REPRESENTED BY A	MOU	INT IN ROW (11)	
	19.0% – See Item 5			
14	TYPE OF REPORTING PERSON			
	со			

## Item 1. Security and Issuer.

The class of equity securities to which this statement relates is the common stock, \$0.01 par value per share (the "Shares"), of Artesyn Technologies, Inc., a Florida corporation ("Artesyn"). The principal executive offices of Artesyn are located at 7900 Glades Rd., Suite 500, Boca Raton, Florida 33434.

## Item 2. Identity and Background.

(a)-(c), (f) The name of the person filing this statement is Emerson Electric Co., a Missouri corporation ("Emerson"). The address of the principal business and the principal office of Emerson is 8000 W. Florissant Avenue, P.O. Box 4100, St. Louis, MO 63136. The name, business address, present principal occupation or employment and citizenship of each director and executive officer of Emerson is set forth on Schedule A and incorporated herein by reference.

Emerson is a global company that brings together technology and engineering to provide innovative solutions for its customers in a wide range of industrial, commercial and consumer markets.

(d)-(e) During the last five years, none of Emerson and, to the best of Emerson's knowledge, any of the persons listed on Schedule A attached hereto has been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or has been a party to a civil proceeding of a judicial or administrative

body of competent jurisdiction and as a result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

#### Item 3. Source and Amount of Funds or Other Consideration.

Emerson has entered into the Voting Agreements described in response to Item 4 in connection with entering into the Merger Agreement described in response to Item 4. Emerson has not given any additional monetary consideration in connection with the Voting Agreements.

#### Item 4. Purpose of Transaction.

(a)-(b) On February 1, 2006, Emerson entered into an Agreement and Plan of Merger (the "Merger Agreement") with Artesyn. Pursuant to the Merger Agreement, a newly formed subsidiary of Emerson will merge into Artesyn (the "Merger"), with Artesyn continuing as the surviving corporation. At the effective time of the Merger, each share of Artesyn common stock issued and outstanding immediately prior to the effective time of the Merger will be converted into the right to receive \$11.00 in cash. The Merger is subject to customary conditions, including approval and adoption of the Merger Agreement by the shareholders of Artesyn.

In connection with the Merger, JANA Partners, LLC ("JANA"), all directors of Artesyn holding shares of Artesyn common stock or options to purchase shares of Artesyn common stock, and certain officers of Artesyn have entered into voting agreements with Emerson dated as of February 1, 2006 (collectively, the "Voting Agreements"). A copy of the Voting Agreement entered into by JANA is attached hereto as Exhibit 99.2. Copies of the Voting Agreements entered into by Artesyn's officers and directors are attached hereto as Exhibits 99.3 -99.13. Pursuant to the Voting Agreements, such shareholders have agreed, among other things, to vote, and have granted Emerson an irrevocable proxy until termination of the Voting Agreements to vote (i) to approve and adopt the Merger Agreement and the Merger, (ii) against any Acquisition Proposal (as defined in the Merger Agreement), reorganization, recapitalization, liquidation or

3

winding-up of Artesyn or any other extraordinary transaction involving Artesyn and (iii) against any corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of, the transactions contemplated by the Merger Agreement. The Voting Agreements with the directors and officers of Artesyn terminate upon the earlier to occur of the Effective Time (as defined in the Merger Agreement) and termination of the Merger Agreement. The Voting Agreement with JANA terminates upon the earliest to occur of (i) the termination of the Merger Agreement, (ii) the occurrence of an Adverse Recommendation Change (as defined in the Merger Agreement) and (iii) a reduction in the merger consideration without JANA's consent or an amendment of the Merger Agreement that materially and adversely affects JANA without JANA's consent.

- (c) Not applicable.
- (d) It is anticipated that upon consummation of the Merger, the officers and directors of Emerson's merger subsidiary at the effective time of the Merger shall become officers and directors of the surviving corporation, until their respective successors are duly elected or appointed and qualified in accordance with applicable law.
  - (e) Other than as a result of the Merger described in this Item, not applicable.
  - (f) Not applicable.
- (g) Upon consummation of the Merger, the Certificate of Incorporation and the Bylaws of Emerson's merger subsidiary as of the effective time of the Merger will become the Certificate of Incorporation and the Bylaws of the surviving corporation.
- (h)-(i) Upon completion of the Merger, Artesyn's Shares will cease to be quoted on the NASDAQ National Market System and will become eligible for termination of registration pursuant to Section 12(g)(4) of the Securities Exchange Act of 1934, as amended.
  - (j) Other than as described above, not applicable.

Reference to, and descriptions of, the Merger Agreement and the Voting Agreements as set forth above in this Item 4 are qualified in their entirety by reference to the copies of the Merger Agreement and the Voting Agreements included as Exhibits 99.1 and 99.2 – 99.13, respectively.

#### Item 5. Interest in Securities of the Issuer.

- (a) Emerson has acquired certain rights with respect to the voting and disposition of and, for the purpose of Rule 13d-3 promulgated under the Exchange Act, may be deemed to beneficially own 7,939,749 Shares, representing approximately 19.0% of the outstanding Shares of Artesyn (based on the number of Shares outstanding as of December 31, 2005, as represented by the issuer in the Merger Agreement, discussed in Item 4 hereof). Emerson expressly disclaims beneficial ownership of such Shares. Except as set forth in this Item 5(a), none of Emerson and, to the best of its knowledge, any persons named in Schedule A hereto, beneficially own any Shares. The number of Shares set forth in this paragraph includes Shares issuable pursuant to options to purchase 1,454,000 Shares which are currently exercisable or exercisable within 60 days of February 1, 2006.
- (b) Emerson has sole power to vote 7,939,749 Shares pursuant to the Voting Agreements as described in Item 4 hereof. The number of Shares set forth in the preceding sentence includes

Shares issuable pursuant to options to purchase 1,454,000 Shares which are currently exercisable or exercisable within 60 days of February 1, 2006.

- (c) To Emerson's knowledge, no transactions in Artesyn have been effected during the past sixty days by any person named pursuant to Item 2 hereof.
- (d) To Emerson's knowledge, no person other than the shareholders that are parties to the Voting Agreements described in Item 4 hereof has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, such Shares.

### (e) Not applicable.

Reference to, and descriptions of, the Merger Agreement and the Voting Agreements as set forth above in this Item 5 are qualified in their entirety by reference to the copies of the Merger Agreement and the Voting Agreements included as Exhibits 99.1 and 99.2 – 99.13, respectively.

## Item 6. Contracts, Arrangements, Understandings or Relationships With Respect to Securities of the Issuer.

See responses to Items 4 and 5 hereof. Except as described above, to the best knowledge of Emerson, there are no contracts, arrangements, understandings or relationships (legal or otherwise), including, but not limited to, transfer or voting of any of the securities, finder's fees, joint ventures, loan or option arrangements, puts or calls, guarantees of profits, division of profits or loss, or the giving or withholding of proxies, between the persons enumerated in Item 2 hereof, and any other person, with respect to any securities of Artesyn, including any securities pledged or otherwise subject to a contingency the occurrence of which would give another person voting power or investment power over such securities other than standard default and similar provisions contained in loan agreements.

#### Item 7. Material to be Filed as Exhibits.

Description

Exhibit No.

99.1	Agreement and Plan of Merger, dated as of February 1, 2006, by and between Artesyn Technologies, Inc., Emerson Electric Co. and Atlanta Acquisition Sub, Inc.			
99.2	Voting Agreement, dated as of February 1, 2006, between Emerson Electric Co. and JANA Partners LLC.			
99.3	Voting Agreement, dated as of February 1, 2006, between Emerson Electric Co. and Edward S. Croft, III.			
	5			
99.4	Voting Agreement, dated as of February 1, 2006, as amended, between Emerson Electric Co. and Gary R. Larsen.			
99.5	Voting Agreement, dated as of February 1, 2006, as amended, between Emerson Electric Co. and Lawrence J. Matthews.			
99.6	Voting Agreement, dated as of February 1, 2006, as amended, between Emerson Electric Co. and Joseph M. O'Donnell.			
99.7	Voting Agreement, dated as of February 1, 2006, between Emerson Electric Co. and Stephen A. Ollendorf.			
99.8	Voting Agreement, dated as of February 1, 2006, between Emerson Electric Co. and Phillip A. O'Reilly.			
99.9	Voting Agreement, dated as of February 1, 2006, as amended, between Emerson Electric Co. and Bert Sager.			
99.10	Voting Agreement, dated as of February 1, 2006, between Emerson Electric Co. and Eugene Sapp, Jr.			
99.11	Voting Agreement, dated as of February 1, 2006, as amended, between Emerson Electric Co. and Ronald D. Schmidt.			
99.12	Voting Agreement, dated as of February 1, 2006, between Emerson Electric Co. and Lewis Solomon.			
99.13	Voting Agreement, dated as of February 1, 2006, as amended, between Emerson Electric Co. and John M. Steel.			
	6			

## SIGNATURE

After reasonable inquiry and to the best of my knowledge and belief, I certify that the information set forth in this statement is true, complete and correct.

EMERSON ELECTRIC CO.

Date: February 13, 2006 By: /s/ Harley M. Smith

Name: Harley M. Smith

Assistant General Counsel and Assistant Secretary

7

Title:

### SCHEDULE A

## DIRECTORS AND EXECUTIVE OFFICERS OF EMERSON ELECTRIC CO.

The name, business address, title, present principal occupation or employment of each of the directors and executive officers of Emerson Electric Co. ("Emerson"), are set forth below. If no business address is given the director's or officer's business address is c/o Emerson Electric Co., 8000 W. Florissant Avenue, P.O. Box 4100, St. Louis, MO 63136. Unless otherwise indicated, each occupation set forth opposite an individual's name refers to Emerson. Unless otherwise indicated below are citizens of the United States of America.

Name and Business Address	Present Principal Occupation Including Name and Address of Employer
Directors A. A. Busch III	Chairman of the Board of Anheuser-Busch Companies, Inc. (brewery, container manufacturer and theme park operator)
D. N. Farr	Chairman of the Board, Chief Executive Officer and President of Emerson
D. C. Farrell	Retired Chairman and Chief Executive Officer of The May Department Stores Company (department stores)
C. Fernandez G.	Chairman and Chief Executive Officer of Grupo Modelo, S.A. de C. V. (brewery) (citizen of the Republic of Mexico)
W. J. Galvin	Senior Executive Vice President and Chief Financial Officer of Emerson
A. F. Golden	Partner of Davis Polk & Wardwell (law firm)
R. B. Horton	Executive Chairman of the Sporting Exchange, Ltd. (online betting exchange), Retired Chairman of BP p.l.c. and Railtrack Group PLC and Former Chairman of Chubb plc (citizen of the United Kingdom)
G. A. Lodge	Retired general partner of InnoCal venture funds (venture capital firm)
V. R. Loucks, Jr	Retired Chairman and Chief Executive Officer of
	Baxter International Inc. (medical products and services)
J. B. Menzer	Vice Chairman of Wal-Mart Stores, Inc. (department stores)
C. A. Peters	Senior Executive Vice President of Emerson
J. W. Prueher	Retired Admiral, U.S. Navy, and Former U.S. Ambassador to The People's Republic of China
R. L. Ridgway	Former Assistant Secretary of State for Europe and Canada
Executive Officers (Who Are Not Directors)	
E. L. Monser	Chief Operating Officer
R. J. Schlueter	Vice President and Chief Accounting Officer
W. W. Withers	Executive Vice President, Secretary and General Counsel of Emerson

## AGREEMENT AND PLAN OF MERGER

by and between

## ARTESYN TECHNOLOGIES, INC.,

## EMERSON ELECTRIC CO., and

## ATLANTA ACQUISITION SUB, INC.

Dated as of: February 1, 2006

## **Table of Contents**

		Page
ARTICLE I	THE MERGER	2
1.1	The Merger	2
1.2	Effective Time	2
ARTICLE II	THE SURVIVING CORPORATION	2
2.1	Effects of the Merger	2
2.2	Articles of Incorporation	2
2.3	By-Laws	3
2.4	Directors	3
2.5	Officers	3
ARTICLE III	CONVERSION OF SECURITIES; EXCHANGE OF CERTIFICATES	3
3.1	Effects on Capital Stock	3
3.2	Exchange of Certificates for Cash	4
3.3	Adjustments to Prevent Dilution	6
3.4	Finestar Warrant	6
3.5	Convertible Notes	7
ARTICLE IV	REPRESENTATIONS AND WARRANTIES	7
4.1	Representations and Warranties of the Company	7
4.2	Representations and Warranties of Parent and Merger Sub	25
ARTICLE V	CONDUCT OF BUSINESS PENDING THE MERGER	27
5.1	Covenants of the Company	27
5.2	Additional Covenants of the Company	29
ARTICLE VI	ADDITIONAL AGREEMENTS	29
6.1	Access	29
6.2	Acquisition Proposals	30
6.3	Stockholders Meeting	32
6.4	Filings; Other Actions; Notification	33
6.5	Vote of Shares Owned by Parent	34
6.6	Publicity	34
6.7	Employee Benefits Matters; Bonus Plan	34
6.8	Indemnification; Directors' and Officers' Insurance; Exculpation	35
6.9	Executive Incentive Plan	36
6.10	Transfer Tax	36
ARTICLE VI	I CONDITIONS	36
7.1	Conditions to Each Party's Obligation to Effect the Merger	36
7.2	Conditions to Parent's and Merger Sub's Obligation to Effect the Merger	37
7.3	Conditions to the Company's Obligation to Effect the Merger	38

i

ARTICLE	VIII TERMINATION	38
8.1	Termination	38
8.2	Effects of Termination	39
8.3	Fees and Expenses	40
ARTICLE I	IX MISCELLANEOUS AND GENERAL	41
9.1	Non-Survival of Representations, Warranties and Agreements	41
9.2	Modification or Amendment	42
9.3	Waiver of Conditions	42
9.4	Counterparts	42
9.5	Governing Law and Venue	42
9.6	WAIVER OF JURY TRIAL	42
9.7	Notices	42
9.8	Entire Agreement; No Other Representations	43
9.9	Binding Effect; No Third-Party Beneficiaries	44
9.10	Obligations of Parent and the Company	44
9.11	Severability	44
9.12	Disclosure Schedules	44
9.13	Interpretation	45
9.14	Assignment	45

## COMPANY DISCLOSURE SCHEDULES

Schedule 4.1(a)(ii) Company Subsidiaries Capital Structure Schedule 4.1(b) Governmental Filings Schedule 4.1(d)(i)(C) Non-Contravention Schedule 4.1(d)(ii)(B) Schedule 4.1(e)(v) Controls Schedule 4.1(e)(vi) Knowledge of the Company Undisclosed Material Liabilities Schedule 4.1(f) Schedule 4.1(g) Absence of Certain Changes or Events Schedule 4.1(h) Litigation Schedule 4.1(i) **Employee Benefits** Schedule 4.1(i)(x) 2006 Executive Incentive Plan Schedule 4.1(j) Compliance with Laws Schedule 4.1(1) **Environmental Matters** Schedule 4.1(m) Tax Matters Schedule 4.1(o)) Labor Matters Schedule 4.1(p) Intellectual Property Schedule 4.1(r) Material Contracts Schedule 4.1(s) Insurance Schedule 5.1 Covenants of the Company Schedule 6.7(d) Employment, Severance and Change of Control Agreements

ii

## PARENT DISCLOSURE SCHEDULES

Schedule 4.2(c)(i) Parent Required Statutory Approvals
Schedule 4.1(e)(vi) Knowledge of Parent

iii

Defined Term	Section
Acquisition Proposal	6.2(a)
Adverse Recommendation Change	6.2(a)
Affected Employees	6.7(a)
Affiliate	4.1(i)(ix)
Agreement	Forepart
Articles of Incorporation	2.2
Articles of Merger	1.2
Bankruptcy and Equity Exception	4.1(c)
Benefit Plans	6.7(b)
Business Day	1.2
By-Laws	2.3
Certificates	3.2(b)
Closing	1.2
Closing Date	1.2
Code	4.1(i)(ii)
Company	Forepart
Company Board Recommendation	4.1(c)
Company Common Stock	3.1(a)
Company Compensation and Benefit Plan	4.1(i)(i)
Company Disclosure Schedules	4.1(a)(ii)
Company Intellectual Property Rights	4.1(p)(i)
Company Material Adverse Effect	4.1(a)(i)
Company Option	3.1(d)(ii)
Company Preferred Stock	4.1(b)
Company Proxy Statement	4.1(w)
Company Reports	4.1(e)(i)
Company Required Statutory Approvals	4.1(d)(i)
Company Requisite Vote	4.1(t)
Company Stock Option Plans	3.1(d)(i)(B)
Company Stockholders Meeting	6.3
Confidentiality Agreement	9.8
Contracts	4.1(d)(ii)
Convertible Notes	3.5
Economic Sanctions	4.1(x)
Effective Time	1.2
Environmental Law	4.1(l)
ERISA	4.1(i)(i)
Exchange Act	4.1(d)(i)
Exchange Agent	3.2(a)
Exchange Fund	3.2(a)
Executive Incentive Plan	4.1(i)(x)
Expenses	8.3(a)
FBCA	Recitals
Finestar Warrant	3.4
Governmental Entity	4.1(d)(i)
Government Official	4.1(x)
HSR Act	4.1(d)(i)
instruction in the second of t	7.1(u)(1)

**Defined Term** Section Indemnified Parties 6.8(a) Indenture 3.5 internal controls 4.1(e)(iv) IRS 4.1(i)(ii) International Plan 4.1(i)(viii) knowledge 4.1(e)(vi) Law(s) 1.2 Lien(s) 4.1(q) Litigation Claims 4.1(h)

Material Contracts4.1(r)MergerRecitalsMerger Consideration3.1(a)Merger SubForepart

iv

Nasdaq	4.1(d)(i)
Parent	Forepart
Parent Disclosure Schedules	4.1(e)(vi)
Parent Material Adverse Effect	4.2(c)
Parent Required Statutory Approvals	4.2(c)(i)
Pension Plan	4.1(i)(ii)
Per Share Amount	3.1(a)
Person	3.1
Proceeding	5.1(m)
Representatives	6.2(a)
Rights Agreement	4.1(b)
SEC	4.1(e)(i)
Securities Act	4.1(e)(i)
Shares	3.1(a)
SOX	4.1(e)(iv)
Subsidiary	4.1(a)(i)
Superior Proposal	6.2(b)
Surviving Corporation	1.1
Tax(es)	4.1(m)(x)
Tax Return	4.1(m)(x)
Termination Date	8.1(b)(i)
Termination Fee	8.3(b)
Transfer Taxes	6.10
Treasury Regulations	4.1(m)(iii)
Trustee	3.5
U.S. GAAP	4.1(a)(i)
Voting Debt	4.1(b)
WARN Act	4.1(o)(iii)

V

## AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of February 1, 2006, by and between ARTESYN TECHNOLOGIES, INC., a Florida corporation (the "Company"), EMERSON ELECTRIC CO., a Missouri corporation ("Parent"), and ATLANTA ACQUISITION SUB, INC., a Florida corporation and a wholly owned subsidiary of Parent ("Merger Sub").

### WITNESSETH:

WHEREAS, the Company has actively pursued a business combination and has engaged a financial advisor to facilitate and manage a sale, merger or other business combination of the Company;

WHEREAS, the Company has considered and reviewed proposals from other third parties for business combinations and desires to proceed with a business combination between the Company, Parent and Merger Sub;

WHEREAS, the respective Boards of Directors of Parent, Merger Sub and the Company have each approved and declared advisable this Agreement and the merger (the "Merger") of Merger Sub with and into the Company upon the terms and subject to the conditions of this Agreement and in accordance with the Florida Business Corporation Act (the "FBCA");

WHEREAS, the Board of Directors of the Company has unanimously determined to recommend to its stockholders the approval and adoption of this Agreement and the Merger and the other transactions contemplated hereby, upon the terms and subject to the conditions hereof;

WHEREAS, Parent, Merger Sub and the Company desire to make certain representations, warranties and agreements in connection with the Merger and the other transactions contemplated by this Agreement and to prescribe certain conditions to the Merger and the other transactions contemplated hereby; and

WHEREAS, concurrently with the execution of this Agreement, as a condition of the willingness of Parent to enter into this Agreement, certain stockholders of the Company have entered into voting agreements providing for, among other things, the agreement of such stockholders to vote their Shares in favor of the Merger and the approval and adoption of this Agreement.

NOW, THEREFORE, in consideration of the premises, and of the mutual representations, warranties, covenants and agreements contained herein, and intending to be legally bound hereby, the parties hereto agree as follows:

1

#### THE MERGER

- 1.1 The Merger. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the FBCA, at the Effective Time: (i) Merger Sub shall be merged with and into the Company; (ii) the separate corporate existence of Merger Sub shall thereupon cease; and (iii) the Company shall be the surviving corporation in the Merger (the "Surviving Corporation") and shall succeed to and assume all the rights, privileges and obligations of Merger Sub.
- 1.2 Effective Time. The closing of the Merger (the "Closing") shall take place at (a) the offices of Kirkpatrick & Lockhart Nicholson Graham LLP, 599 Lexington Avenue, New York, New York, at 10:00 a.m. Eastern Time, as soon as practicable, and in no event later than five Business Days, after the satisfaction of each of the conditions set forth in Article VII (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction of those conditions) or (b) such other place and time and/or on such other date as the Company and Parent may agree in writing (either clause (a) or (b), the "Closing Date"). At the Closing, the parties hereto shall cause the Merger to be consummated by filing articles of merger (the "Articles of Merger"), setting forth this Agreement or an appropriate plan of merger with the Secretary of State of the State of Florida, in such form as is required by, and executed in accordance with, the relevant provisions of the FBCA (the date and time of such filing being the "Effective Time"). As used herein, the term "Business Day" shall mean any day other than a Saturday, Sunday, federal holiday or other day on which banks are required or authorized by Law to close in New York City. As used herein, the term "Laws" shall mean any U.S. (federal, state or local) or foreign law (including common law), statute, rule, regulation, order, judgment or decree.

## ARTICLE II

### THE SURVIVING CORPORATION

- 2.1 Effects of the Merger. At the Effective Time, the effects of the Merger shall be as provided in this Agreement, the Articles of Merger and the applicable provisions of the FBCA. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all of the properties, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and duties of the Surviving Corporation.
- 2.2 Articles of Incorporation. At the Effective Time, the articles of incorporation of Merger Sub, as in effect immediately prior to the Effective Time, shall be the articles of incorporation of the Surviving Corporation until thereafter amended as provided therein or by applicable Law (the "Articles of Incorporation"), except that Article I thereof shall be amended, by the filing of the Articles of Merger or other appropriate documents, to read in its entirety as follows: "The name of the corporation is Artesyn Technologies, Inc."

2

- 2.3 <u>By-Laws</u>. At the Effective Time, and without any further action on the part of the Company or Merger Sub, the by-laws of Merger Sub, as in effect immediately prior to the Effective Time and including the provisions required by Section 6.8(c), shall be the by-laws of the Surviving Corporation (the "<u>By-Laws</u>") until, subject to Section 6.8(c), thereafter amended as provided therein, in the Articles of Incorporation or by applicable Law.
- 2.4 <u>Directors</u>. From and after the Effective Time, the directors of Merger Sub at the Effective Time shall be the directors of the Surviving Corporation until their successors shall have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and the By-Laws.
- 2.5 Officers. From and after the Effective Time, the officers of Merger Sub at the Effective Time shall be the officers of the Surviving Corporation until their successors shall have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and the By-Laws.

#### ARTICLE III

## CONVERSION OF SECURITIES; EXCHANGE OF CERTIFICATES

- 3.1 Effects on Capital Stock. At the Effective Time, as a result of the Merger and without any further action on the part of the Company, Parent, Merger Sub or any holder of any capital stock of such Persons:
- (a) <u>Conversion of Securities</u>. Each share of the Company's common stock, par value \$0.01 per share (the "<u>Company Common Stock</u>"), issued and outstanding immediately prior to the Effective Time (collectively, the "<u>Shares</u>") (other than Shares to be canceled pursuant to Section 3.1(b)) shall no longer be outstanding and shall automatically be canceled and converted into the right to receive \$11.00 per Share in cash, without interest (the "<u>Per Share Amount</u>") (the aggregate cash amount payable pursuant to this Section 3.1(a) being hereinafter referred to as the "<u>Merger Consideration</u>").
- (b) <u>Cancellation of Treasury Stock and Parent Owned Stock</u>. Each share of Company Common Stock held in the treasury of the Company and any Shares held immediately prior to the Effective Time by the Company, Parent or any Subsidiary of the Company or of Parent (including Merger Sub) shall automatically be retired and shall cease to be outstanding, and no Merger Consideration shall be delivered or deliverable in exchange therefor or in consideration thereof.
- (c) <u>Conversion of Merger Sub Stock</u>. Each share of common stock of Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one validly issued, fully-paid and non-assessable share of common stock of the Surviving Corporation.

3

## (d) Company Options, Etc.

(i) Prior to or simultaneously with the execution of this Agreement, the Board of Directors of the Company and the Compensation and Stock Option Committee of the Board have adopted resolutions with respect to the Company's 1990 Performance Equity Plan, 2000 Performance Equity Plan and 1990 Outside Director Stock Option Plan and any other Company Compensation and Benefit Plan under which any option or other right to acquire shares of Company Common Stock has been granted (collectively, the "Company Stock Option Plans") to implement the provisions of Section 3.1(d)(ii).

- (ii) At the Effective Time, each outstanding option or other right to acquire shares of Company Common Stock, whether or not presently exercisable, (collectively, the "Company Options") granted under the Company Stock Option Plans shall cease to be a right to purchase shares of Company Common Stock and shall become a right to receive an amount in cash equal to the product of (1) the excess, if any, of the Per Share Amount over the exercise price per share of the Company Common Stock subject to such Company Option and (2) the number of shares of Company Common Stock issuable pursuant to the unexercised portion of such Company Option; and the Company Stock Option Plans shall terminate, and no Person shall thereafter be granted any rights under such Plans.
- (iii) Prior to the Effective Time, the Company (A) shall use reasonable best efforts to take such actions as shall be necessary or desirable to provide that the Surviving Corporation shall be entitled to any applicable federal or state income tax deductions attributable to the payments to holders of Company Options as provided above, and (B) shall cooperate with Parent and Merger Sub in providing notices and other communications to holders of Company Options regarding the settlement of the Company Options as provided above.
- (iv) All amounts payable pursuant to this Section 3.1(d) shall be subject to any required withholding of Taxes and shall be paid at or as soon as practicable following the Effective Time, but in any event within seven days following the Effective Time, without interest.

As used herein, the term "Person" shall mean any individual, corporation (including not-for-profit corporation), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization or other entity of any kind or nature.

### 3.2 Exchange of Certificates for Cash.

(a) Exchange Agent. As of the Effective Time, Parent shall deposit, or shall cause to be deposited, with an exchange agent reasonably acceptable to the Company (the "Exchange Agent"), for the benefit of the holders of Shares and holders of Company Options and for exchange in accordance with this Article III through the Exchange Agent, the Merger Consideration and the aggregate consideration to which holders of Company Options shall become entitled pursuant to Section 3.1(d) (such Merger Consideration and consideration relating to Company Options, together with any interest earned thereon, being hereinafter

4

referred to as the "Exchange Fund"). The Exchange Agent shall, pursuant to irrevocable instructions to be given by Parent, deliver the cash (excluding any interest earned thereon) out of the Exchange Fund. The Exchange Fund shall not be used for any purpose that is not expressly provided for in this Agreement; provided, that Parent may direct the Exchange Agent to invest the Exchange Fund in obligations of or guaranteed by the United States of America and backed by the full faith and credit of the United States of America or in commercial paper obligations rated P-1 or A-1 or better by Moody's Investors Services, Inc. or Standard & Poor's Corporation, respectively. Any interest or other income resulting from such investments shall be promptly paid to Parent.

- (b) Exchange Procedures. As promptly as practicable after the Effective Time, Parent shall cause the Exchange Agent to mail to each holder of a certificate or certificates that immediately prior to the Effective Time represented Shares (the "Certificates"): (i) a letter of transmittal (which letter shall specify that delivery shall be effected, and risk of loss and title to such Shares shall pass, only upon proper delivery of Certificates to the Exchange Agent and shall be in customary form); and (ii) instructions for effecting the surrender of the Certificates in exchange for the portion of the Merger Consideration payable to such holder in accordance with Section 3.1(a). Upon surrender to the Exchange Agent of a Certificate for cancellation, together with such letter of transmittal, duly executed, and such other documents as may be required pursuant to such instructions, the holder of such Certificate shall be entitled to receive, in exchange therefor, the portion of the Merger Consideration payable to such holder in accordance with Section 3.1(a) (after giving effect to any required Tax withholdings) in respect of the Shares formerly represented by such Certificate, and the Certificate so surrendered shall forthwith be canceled. No interest will be paid or will accrue on the portion of the Merger Consideration payable upon surrender of any Certificate. In the event of a transfer of ownership of Shares that is not registered in the transfer records of the Company, such portion of the Merger Consideration may be paid to the transfere if the Certificate representing such Shares is presented to the Exchange Agent, accompanied by all documents reasonably required to evidence and effect such transfer and by evidence that any applicable stock transfer Taxes have been paid.
- (c) Stock Transfer Books. As of the Effective Time, the stock transfer books of the Company shall be closed, and there shall be no further registration of transfers of the Shares thereafter on the records of the Company or the Surviving Corporation. From and after the Effective Time, the holders of Certificates shall cease to have any rights with respect to the Shares represented thereby, except as otherwise provided herein or by applicable Law. On or after the Effective Time, any Certificates duly presented to the Exchange Agent or Parent for any reason shall be exchanged as provided in Section 3.2(b).
- (d) <u>Termination of Exchange Fund</u>. Any portion of the Exchange Fund that remains undistributed to the holders of Shares outstanding immediately prior to the Effective Time one (1) year after the Effective Time shall be delivered to Parent, upon its written demand therefor, and any such holders that have not theretofore exchanged Certificates pursuant to this Article III shall thereafter look only to Parent for payment of the portion of the Merger Consideration to which they are entitled. Any portion of the Exchange Fund remaining unclaimed by such holders as of the date that is immediately prior to such date that such amount would otherwise escheat to or become the property of any Governmental Entity shall, if and to

5

the extent permitted by applicable Law, be unconditionally and irrevocably delivered to the Surviving Corporation, free and clear of any claims or interests of any Person previously entitled thereto. Notwithstanding the foregoing, none of Parent, the Surviving Corporation, the Exchange Agent or any other Person shall be liable to any former holder of the Shares for any amount properly delivered to a public official as required pursuant to applicable abandoned property, escheat or similar Laws.

- (e) <u>Lost, Stolen or Destroyed Certificates</u>. In the event that any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if reasonably required by Parent, the posting by such Person of a bond in customary amount as indemnity against any claim that may be made against it or the Surviving Corporation with respect to such Certificate, the Exchange Agent will issue, in exchange for such lost, stolen or destroyed Certificate, the portion of the Merger Consideration payable in respect of the Shares formerly represented thereby.
- (f) Withholding Rights. Each of Parent, the Surviving Corporation and the Exchange Agent may deduct and withhold from the portion of the Merger Consideration otherwise payable to any holder of Shares pursuant to this Article III such amounts, without duplication, as it is required to deduct and withhold with respect to the making of such payment

under any applicable Tax Law. If Parent, the Surviving Corporation or the Exchange Agent, as the case may be, so deducts and withholds such amounts, such amounts shall be treated for all purposes of this Agreement as having been paid to the holder of the Shares in respect of which Parent, the Surviving Corporation or the Exchange Agent, as the case may be, shall make such deduction and withholding.

- 3.3 Adjustments to Prevent Dilution. Notwithstanding anything to the contrary contained in this Agreement, if after the date hereof and prior to the Effective Time, the issued and outstanding Shares shall have been changed into a different number of shares or a different class by reason of any stock split, reverse stock split, stock dividend, reclassification, recapitalization, split-up, combination, exchange of shares or similar transaction, then the Merger Consideration and any other similarly dependent items, as the case may be, shall be appropriately adjusted to provide to the holders of Shares the same economic effect as contemplated by this Agreement prior to such action, and as so adjusted shall, from and after the date of such event, be the Merger Consideration, or such other dependent items, as applicable, subject to further adjustment in accordance with this Section 3.3.
- 3.4 Finestar Warrant. As of the Effective Time, if the Finestar Warrant shall not have been previously exercised in full, the Surviving Corporation shall, in accordance with the terms of the Finestar Warrant, enter into a supplemental agreement providing that the holder of the Finestar Warrant shall be entitled to receive, and the Finestar Warrant shall thereafter represent the right to receive (upon due exercise and payment of the exercise price), in lieu of the shares of Company Common Stock issuable upon exercise of the Finestar Warrant immediately prior to the Effective Date, the amount of Merger Consideration payable pursuant to the Merger in respect of such number of shares of Company Common Stock that is equal to the number of shares of Company Common Stock into which the Finestar Warrant would otherwise have been exercisable absent consummation of the Merger. As used herein, the term "Finestar Warrant"

6

shall mean Warrant No. CS-1 of the Company issued on January 15, 2002 to Finestar International Limited.

3.5 Convertible Notes. As of the Effective Time, the Surviving Corporation, shall, in accordance with the Indenture, enter into a supplemental indenture (which indenture shall comply with the Trust Indenture Act of 1939 as in effect at the date of execution of such supplemental indenture) providing that the Convertible Notes shall be convertible into, in lieu of the shares of Company Common Stock issuable upon conversion of the Convertible Notes immediately prior to the Effective Date, the amount of Merger Consideration payable pursuant to the Merger in respect of such number of shares of Company Common Stock that is equal to the number of shares of Company Common Stock into which the Convertible Notes would otherwise have been convertible absent consummation of the Merger. Promptly after the Effective Time, Parent shall, or shall cause the Surviving Corporation to, deliver to the holders of the Convertible Notes the notices required by the terms of the Indenture as a result of the transactions contemplated hereby, and shall comply with the terms of the Indenture with respect to any repurchase of Convertible Notes from the holders thereof as a result of the transactions contemplated hereby, as required by the terms of such Indenture. As used herein, the terms (i) "Indenture" shall mean the Indenture, dated as of August 13, 2003, between the Company and The Bank of New York, as trustee (the "Trustee"), relating to the Company's 5.50% Convertible Senior Subordinated Notes Due 2010 and (ii) "Convertible Notes" shall mean the convertible notes referred to in clause (i) above. The Company shall obtain all opinions, certificates and other documents required to be provided by the Company or its counsel pursuant to the Indenture in connection with the execution of such supplemental indenture.

#### ARTICLE IV

## REPRESENTATIONS AND WARRANTIES

- 4.1 Representations and Warranties of the Company. The Company hereby represents and warrants to Parent and Merger Sub that:
- (a) Organization, Good Standing and Qualification; Organizational Documents.
- (i) The Company and each of its Subsidiaries is an entity duly organized, validly existing and in good standing (with respect to jurisdictions that recognize such concept) under the Laws of its respective jurisdiction of organization and has all requisite corporate (or similar) power and authority to own, license and operate its properties and assets and to carry on its business as presently conducted, except where the failure to be so organized, existing or in good standing or to have such power and authority has not had, and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Each of the Company and its Subsidiaries is duly qualified or licensed to do business and is in good standing in each jurisdiction where the ownership, leasing or operation of its properties and assets or the conduct of its business requires such qualification or licensure, except where the failure to be so qualified or licensed or in good standing has not had, and could

7

not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

As used herein, the term "<u>Subsidiary</u>" shall mean, with respect to any particular Person, any entity, whether incorporated or unincorporated, of which a majority of the outstanding securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other Persons performing similar functions is directly or indirectly owned or controlled by such Person or by one or more of its Subsidiaries or by such Person and any one or more of its Subsidiaries.

As used herein, the term "Company Material Adverse Effect" shall mean a material adverse effect on the business, operations, assets, liabilities, financial condition or results of operations of the Company and its Subsidiaries, taken as a whole; provided, however, that none of the following shall be deemed in and of themselves, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been a Company Material Adverse Effect: (i) any changes in United States generally accepted accounting principles ("U.S. GAAP"); (ii) changes in general economic conditions in the United States or the global economy; (iii) changes (including changes in Law) affecting the industry in which the Company operates in general, to the extent the Company and its Subsidiaries, taken as a whole, are not disproportionately affected thereby; (iv) any adverse change, effect, event, occurrence, state of facts or development (including any cancellations of or delays in customer orders, any reduction in sales or results of operations, any disruption in supplier, distributor, partner or similar relationships or any loss of employees) attributable to the announcement or pendency of the Merger; or (v) any action required to be taken pursuant to this Agreement or by or at the request of Parent or Merger Sub. The parties agree that the mere fact of (A) a decrease in the market price or an increase or decrease in the trading volume of the Company Common Stock or (B) the failure of the Company to meet internal projections or published revenues or earnings forecasts shall not by itself constitute a Company Material Adverse Effect, but the reasons for or causes of such events (to the extent not due to the announcement or pendency of the Merger) may constitute a Company Material Adverse Effect.

(ii) The Company has heretofore made available to Parent true and correct copies of the Company's articles of incorporation and by-laws, each as amended through the date hereof. Schedule 4.1(a)(ii) of the disclosure schedules delivered to Parent by the Company on or prior to the date hereof (the "Company Disclosure Schedules") sets forth a true and correct list of all Subsidiaries of the Company, the jurisdictions in which such Subsidiaries are organized and the percent of the equity interest therein owned by the Company and any other Subsidiary of the Company or by any other Person, as applicable.

(b) <u>Capital Structure</u>. The authorized capital stock of the Company consists of 80,000,000 shares of Company Common Stock and 1,000,000 shares of preferred stock, par value \$0.01 (the "<u>Company Preferred Stock</u>"). As of the close of business on December 31, 2005, there were outstanding (i) 40,244,326 shares of Company Common Stock (including 279,575 shares subject to restrictions under the Company Stock Option Plans), (ii) no shares of Company Preferred Stock, (iii) Company Options to purchase an aggregate of 4,897,492 shares of Company Common Stock at a weighted average exercise price of \$9.41 per share, (iv) the Finestar Warrant to purchase an aggregate of 1,550,000 shares of Company Common Stock at an

8

exercise price of \$10.73 per share and (v) \$90 million principal amount of Convertible Notes then convertible into an aggregate of 11,160,714 shares of Company Common Stock. All of the issued and outstanding shares of Company Common Stock have been duly authorized and are validly issued, fully paid and nonassessable and all of the shares of Company Common Stock that may be issued pursuant to the Company Stock Option Plans, the Convertible Notes and the Finestar Warrant have been duly authorized and will be, when issued in accordance with the respective terms thereof, validly issued, fully paid and nonassessable. Except as set forth on Schedule 4.1(b) of the Company Disclosure Schedules, all of the issued and outstanding shares of capital stock of each of the Company's Subsidiaries have been duly authorized and are validly issued, fully paid and nonassessable, and all such shares are owned by the Company or a Subsidiary of the Company free and clear of any Lien (excluding any securities Law general restrictions on transfer). Except as set forth in the second sentence of this Section 4.1(b) and for changes since December 31, 2005 resulting from any exercise of Company Options outstanding as of December 31, 2005, conversion of the Convertible Notes or exercise of the Finestar Warrant, there are no outstanding shares of capital stock or voting securities of the Company. Except as set forth on Schedule 4.1(b) of the Company Disclosure Schedules and other than pursuant to (i) the Company's Rights Agreement, dated as of November 21, 1989, as amended as of October 22, 2004 (the "Rights Agreement"); (ii) the Company Options outstanding as of December 31, 2005; and (iii) the Convertible Notes and the Finestar Warrant, there are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements or commitments to issue or to sell any shares of capital stock or other securities of the Company or any of its Subsidiaries that have been issued or granted by the Company or such Subsidiaries or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any securities of the Company or any of its Subsidiaries. Schedule 4.1(b) of the Company Disclosure Schedules sets forth a true and correct list, as of December 31, 2005, of the Company Options, together with the numbers of shares covered thereby, dates of grant thereof and the exercise prices therefor. Except for the Convertible Notes, the Company does not have any outstanding bonds, debentures, notes or other debt obligations the holders of which have the right to vote (or convertible into or exercisable for securities having the right to vote) with the stockholders of the Company on any matter ("Voting Debt"). No Subsidiary of the Company owns any Company Common Stock.

(c) Corporate Authority. The Company has the requisite corporate power and authority, and has taken all corporate action necessary in order, to execute, deliver and perform its obligations under this Agreement and to consummate, upon the terms and subject to the conditions of this Agreement, the transactions contemplated hereby, except for obtaining the Company Required Statutory Approvals and, with respect to the Merger, the receipt of the Company Requisite Vote. This Agreement has been duly executed and delivered by the Company and, assuming due authorization, execution and delivery by each of Parent and Merger Sub, constitutes the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws of general applicability relating to or affecting creditors' rights and except for general equity principles (the "Bankruptcy and Equity Exception"). At a meeting duly called, convened and held, the Company's Board of Directors unanimously adopted resolutions: (i) approving this Agreement and the transactions contemplated hereby, including the Merger; (ii) declaring that this

9

Agreement is advisable; (iii) stating that the terms of the Merger are fair to, and in the best interests of, the Company and its stockholders; and (iv) recommending that the Company's stockholders approve and adopt this Agreement and the Merger (the "Company Board Recommendation").

#### (d) Governmental Filings; No Violations.

(i) Except for any reports, filings, registrations, approvals and notices required to be made: (A) pursuant to Section 1.2; (B) under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the Nasdaq National Market Stock Market ("Nasdaq"); (C) as set forth on Schedule 4.1(d)(i)(C) of the Company Disclosure Schedules (items (B) and (C), the "Company Required Statutory Approvals"); and (D) by Parent or Merger Sub on behalf of itself and the Company in Italy with the Italian Competition Authority, no notices, reports, registrations or other filings are required to be made by the Company with, and no consents, approvals, permits or authorizations are required to be obtained by the Company from, any U.S. or foreign federal, state or local governmental or regulatory authority, agency, commission, body or other governmental entity (each, a "Governmental Entity"), in connection with the Company's execution and delivery of this Agreement and the consummation by the Company of the Merger and the other transactions contemplated hereby, other than those notices, reports, registrations or other filings and consents, approvals, permits and authorizations the failure of which to make or obtain has not had, and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect or would not prevent, materially delay or materially impair the ability of the Company to consummate the transactions contemplated by this Agreement.

(ii) The execution, delivery and performance of this Agreement by the Company do not, and the consummation by the Company of the Merger and the other transactions contemplated hereby will not (A) constitute or result in a breach or violation of either the articles of incorporation or by-laws of the Company, (B) except as set forth on Schedule 4.1(d)(ii)(B) of the Company Disclosure Schedules, require any consent or other action by any Person under, constitute or result in a breach or violation of, or constitute a default, or give rise to any right of termination, cancellation or acceleration, under the terms of any lease, note, mortgage, license, franchise, indenture, agreement or other contract (collectively, "Contracts") not otherwise terminable on 30 days' or less notice without requiring any payment or incurring any liability to which the Company or any Subsidiary is a party or by which it, or any of its properties or assets, is otherwise bound, (C) assuming compliance with the matters referred to in Section 4.1(d)(i), contravene or conflict with or result in a violation or breach of any Law or (D) result in the creation or imposition of any Lien on any asset of the Company or any of its Subsidiaries, except, in the case of each of clauses (B) through (D) above, for any failure to obtain consent, contravention, conflict, breach, violation, default, termination, cancellation, acceleration or imposition

#### (e) Company Reports; Financial Statements.

- (i) The Company has filed with the Securities and Exchange Commission (the "SEC"), the forms, reports and other documents required to be filed by it since December 31, 2002 under the Securities Act of 1933, as amended (the "Securities Act"), and the Exchange Act (any such forms, reports and other documents, including any amendments of or supplements thereto, filed with or furnished to the SEC by the Company prior to the date hereof, the "Company Reports"). The Company Reports, when filed, were in compliance in all material respects with the requirements of the Securities Act or the Exchange Act, as the case may be, and the rules and regulations thereunder. None of the Company Reports, as of the filing date or, if amended, as of the date the last amendment was filed, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements made therein, in the light of the circumstances under which they were made, not misleading.
- (ii) Each of the consolidated balance sheets included in or incorporated by reference into the Company Reports (including the related notes and schedules thereto) fairly presented, in all material respects, the consolidated financial position of the Company and its Subsidiaries as of its date, and each of the consolidated statements of income (loss) and consolidated statements of cash flows included in or incorporated by reference into the Company Reports (including any related notes and schedules thereto) fairly presented, in all material respects, the consolidated results of operations and cash flows, as the case may be, of the Company and its Subsidiaries for the periods set forth therein (subject, in the case of unaudited statements, to the absence of notes, similar presentation items and normal year-end audit adjustments), in each case in accordance with U.S. GAAP consistently applied during the periods presented, except as may have been noted therein or, in the case of unaudited interim financial statements, as may have been permitted by Form 10-Q under the Exchange Act.
- (iii) The Company has established and maintains disclosure controls and procedures (as defined in Rule 13a-15 under the Exchange Act). Such disclosure controls and procedures are designed to ensure that material information relating to the Company and its Subsidiaries, is made known to the Company's principal executive officer and its principal financial officer by others within those entities. Such disclosure controls and procedures are effective in timely alerting the Company's principal executive officer and principal financial officer to material information required to be included in the Company's periodic reports required under the Exchange Act.
- (iv) The Company and its Subsidiaries have established and maintained a system of internal control over financial reporting (as defined in Rule 13a-15 under the Exchange Act) ("internal controls"). Such internal controls are sufficient to provide reasonable assurance regarding the reliability of the Company's financial reporting and the preparation of the Company's financial statements for external purposes in accordance with U.S. GAAP, including reasonable assurance (i) that receipts and expenditures of the Company are being made only in accordance with authorizations of management and directors of the Company and transactions are recorded as necessary to permit preparation of financial statements in conformity with U.S. GAAP and (ii) regarding prevention or timely detection of any unauthorized acquisition, use or disposition of Company assets that could have a material effect on the

11

Company's financial statements. The Company's management, with the participation of the Company's principal executive and financial officers, has completed its assessment of the effectiveness of the Company's internal controls over financial reporting in compliance with the requirements of Section 404 of the Sarbanes-Oxley Act of 2002 ("SOX") for the year ended December 31, 2004, and such assessment concluded that such internal controls were effective using the framework specified in the Company's Form 10-K filed on March 16, 2005. The Company has disclosed, based on its most recent evaluation of internal controls prior to the date hereof, to the Company's auditors and audit committee (x) any significant deficiencies or material weaknesses in the design or operation of internal controls which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information and (y) any fraud, whether or not material, that involves management or other employees who have a significant role in internal controls. The Company has made available to Parent a summary of any such disclosure made by its management to the Company's auditors and audit committee since December 31, 2002.

- (v) Except as set forth on Schedule 4.1(e)(v) of the Company Disclosure Schedules, since December 31, 2002, (i) neither the Company nor any of its Subsidiaries nor, to the Company's knowledge, any director, officer, employee, auditor, accountant or representative of the Company or any of its Subsidiaries has received any written complaint, allegation, assertion, or claim that the Company or any of its Subsidiaries has engaged in improper or illegal accounting or auditing practices or maintains improper or inadequate internal accounting controls and (ii) no attorney representing the Company or any of its Subsidiaries, whether or not employed by the Company or any of its Subsidiaries, has reported evidence of a material violation of U.S. federal or state securities Laws, a material breach of fiduciary duty or similar material violation by the Company, any of its Subsidiaries or any of their respective officers, director, employees or agents to the chief legal counsel or the chief executive officer of the Company, the Board of Directors of the Company or any member or committee thereof.
- (vi) As used herein, the term " $\underline{knowledge}$ " or any similar formulation shall mean the actual knowledge of, with respect to the Company, only those persons set forth on  $\underline{Schedule\ 4.1(e)(vi)}$  of the Company Disclosure Schedules and, with respect to Parent, only those persons set forth on  $\underline{Schedule\ 4.1(e)(vi)}$  of the disclosure schedules delivered to the Company by Parent on or prior to the date hereof (the " $\underline{Parent\ Disclosure\ Schedules}$ ").
- (vii) There are no outstanding loans or other extensions of credit made by the Company or any of its Subsidiaries to any executive officer (as defined in Rule 3b-7 under the Exchange Act) or director of the Company. The Company has not, since the enactment of SOX, taken any action prohibited by Section 402 of SOX.
- (f) No Undisclosed Material Liabilities. Except as set forth on Schedule 4.1(f) of the Company Disclosure Schedules, as reflected on the consolidated balance sheet of the Company and its Subsidiaries as of September 30, 2005 and as otherwise disclosed in the Company Reports, to the knowledge of the Company, there are no liabilities or obligations of the Company or any of its Subsidiaries, whether accrued, contingent, absolute or otherwise, other than liabilities or obligations: (i) under existing Contracts of the Company and its Subsidiaries that were entered into in the ordinary course of business consistent with past practices and where

neither the Company nor any of its Subsidiaries is in material breach of its obligations thereunder; (ii) incurred in the ordinary course of business consistent with past practice since September 30, 2005; (iii) that have not had, and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect; or (iv) incurred pursuant to this Agreement and in connection with the transactions contemplated hereby.

- (g) Absence of Certain Changes or Events. Except as disclosed in the Company Reports, as contemplated by this Agreement or as set forth on Schedule 4.1(g) of the Company Disclosure Schedules, since September 30, 2005, the Company has conducted its businesses in the ordinary course and in a manner consistent with past practice, and there has not been: (i) any event that has had, or could reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect; (ii) any material change by the Company in its accounting methods, principles or practices except for any such change required by reason of a concurrent change in U.S. GAAP. (iii) any revaluation by the Company of any material asset (including any writing-down of the value of inventory or writing-off of notes or accounts receivable), other than in the ordinary course of business consistent with past practice; (iv) any declaration, setting aside or payment of any dividend or distribution in respect of any capital stock of the Company or any redemption, purchase or other acquisition by the Company or any of its Subsidiaries of any of its securities; (v) any amendment of any material term of any outstanding security of the Company or any of its Subsidiaries; (vi) any settlement or compromise of any material litigation, action or claim; (vii) any establishment or material amendment of any bonus, insurance, severance, deferred compensation, pension, retirement, profit sharing, stock option, stock purchase or other employee benefit plan or any material increase in the compensation payable or to become payable to any officers or key employees of the Company or any of its Subsidiaries, except for salary increases and benefit accruals pursuant to the Company Compensation and Benefit Plans in the ordinary course of business; (viii) any incurrence, assumption or guarantee by the Company or any of its Subsidiaries of any indebtedness for borrowed money other than in the ordinary course of business and in amounts and on terms consistent with past practice; (ix) any creation or other incurrence by the Company or any of its Subsidiaries of any Lien on any material asset other than in the ordinary course of business consistent with past practice; (x) any making of any material loan, advance or capital contributions to or investment in any Person other than loans, advances or capital contributions to or investments in its wholly-owned Subsidiaries in the ordinary course of business consistent with past practice; or (xi) any damage, destruction or other casualty loss (whether or not covered by insurance) affecting the business or assets of the Company or any of its Subsidiaries that has had, or could reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (h) <u>Litigation</u>. Except as disclosed in the Company Reports or as set forth on <u>Schedule 4.1(h)</u> of the Company Disclosure Schedules, there are no civil, criminal or administrative actions, suits, claims, hearings, investigations or proceedings (collectively, "<u>Litigation Claims</u>") pending or, to the knowledge of the Company, threatened against the Company or any of its Subsidiaries or any of their respective properties or, to the knowledge of the Company, pending or threatened against any present or former officer, director or employee of the Company or any of its Subsidiaries before any court or arbitrator or before or by any Governmental Entity that, if adversely determined or resolved, could reasonably be expected to

13

have, individually or in the aggregate, a Company Material Adverse Effect or that seek to prevent or materially delay the consummation of the Merger.

#### (i) Employee Benefits.

- (i) As used herein, the term "Company Compensation and Benefit Plan" shall mean any written or oral bonus, deferred compensation, pension, retirement, profit-sharing, thrift, savings, employee stock ownership, stock bonus, stock purchase, change in control, retention, restricted stock, stock option, employment, termination, severance, compensation, medical, health, employee assistance program, disability or sick leave, workers' compensation, supplemental unemployment benefits, post-employment or retirement benefits (including compensation, pension, health, medical or life insurance benefits) or other compensation or benefit plan, including each "employee benefit plan," within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), that covers current or former employees or directors of the Company or any of its Subsidiaries and is maintained, administered or contributed to by the Company or any of its Subsidiaries or with respect to which the Company or any of its Subsidiaries has any liability. The Company has made available to Parent true and correct copies of all material Company Compensation and Benefit Plans (and, if applicable, related trust or funding agreements or insurance policies).
- (ii) Each of the Company Compensation and Benefit Plans has been maintained in compliance in all material respects with its terms and with the requirements prescribed by any and all applicable Laws (including, if applicable, ERISA and the Internal Revenue Code of 1986, as amended (the "Code")). There has been no amendment to, written interpretation of or announcement (whether or not written) by the Company or any Subsidiary relating to, or change in employee participation or coverage under, any Company Compensation and Benefit Plan that could reasonably be expected to increase materially the expense of maintaining such Company Compensation and Benefit Plan above the level of expense incurred in respect thereof for the most recent fiscal year ended prior to the date hereof. Each Company Compensation and Benefit Plan that is an "employee pension benefit plan" within the meaning of Section 3(2) of ERISA (a "Pension Plan") and that is intended to be qualified under Section 401(a) of the Code has received a favorable determination letter (or opinion letter, if applicable) from the Internal Revenue Service (the "IRS") and, to the Company's knowledge, nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause the loss of such qualification. The Company has provided (or made available to) Parent with the most recent determination letter (or opinion letter, if applicable) of the IRS relating to each such Pension Plan.
- (iii) No Company Compensation and Benefit Plan that is a Pension Plan is subject to the funding rules of Part 3 of Title I of ERISA or is a "multiemployer plan," as such term is defined in Section 3(37) of ERISA.
- (iv) Except as set forth on Schedule 4.1(i)(iv) of the Company Disclosure Schedules, (A) all contributions required to be made by the Company or its Subsidiaries to any Company Compensation and Benefit Plan under the terms of such Plan or under U.S. Law or any other applicable Law have been made or have been accrued on the most recent consolidated balance sheet filed or incorporated by reference in the Company Reports and

- (B) all unfunded benefit obligations of the Company and its Subsidiaries under any Company Compensation and Benefit Plan have been accounted for by reserves or otherwise properly footnoted in accordance with U.S. GAAP on the most recent consolidated balance sheet filed or incorporated by reference in the Company Reports.
- (v) Neither the Company nor its Subsidiaries has any current or projected liability with respect to retiree health and life benefits under any Company Compensation and Benefit Plan subject to U.S. Law, except for benefits required to avoid excise tax under Code Section 4980B or, as set forth on Schedule 4.1(i)(v) of the Company Disclosure Schedules, under any other applicable Law requiring continuation of health coverage.
- (vi) Except as set forth on Schedule 4.1(i)(vi) of the Company Disclosure Schedules: neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereunder will (A) result in an increase in the amount of compensation or benefits or the acceleration of the vesting or timing of payment of any compensation or benefits payable to or in respect of any participant under a Company Compensation and Benefit Plan; or (B) result in or satisfy a condition to the payment of compensation that would, in combination with any other payment, result in an "excess parachute payment" within the meaning of Section 280G(b) of the Code.
- (vii) No transaction prohibited by Section 406 of ERISA or Section 4975 of the Code has occurred with respect to any employee benefit plan or arrangement that is covered by Title I of ERISA, which transaction has or will cause the Company or any of its Subsidiaries to incur any liability under ERISA, the Code or otherwise, excluding transactions effected pursuant to and in compliance with a statutory or administrative exemption and transactions that have not had, and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (viii) The Company and its Subsidiaries have employment policies and practices that are in compliance in all material respects with all currently applicable Laws respecting employment and employment practices, terms and conditions of employment, wages, hours and immigration.
- (ix) Except as set forth on Schedule 4.1(i)(ix) of the Company Disclosure Schedules or as has not had, and could not reasonably be expected to have, a Company Material Adverse Effect, with respect to each Company Compensation and Benefit Plan covering employees outside of the United States (each, an "International Plan"), (A) each International Plan has been maintained in compliance in all material respects with its terms and with the requirements prescribed by any and all applicable statutes, orders, rules and regulations (including any special provisions relating to qualified plans where such Plan was intended to so qualify) and has been maintained in good standing with applicable regulatory authorities and (B) as of September 30, 2005, the sum of (to the extent applicable to a particular Plan) (i) the fair market value of the assets of such Plan to the extent funded, (ii) the liability of each insurer for such Plan to the extent insured, and (iii) the book reserve established for such Plan to the extent neither funded nor insured, together in each case with any accrued contributions, premium payments or benefit accruals, is sufficient to provide for the Projected Benefit Obligation as defined by SFAS 87, with respect to all current and former participants in each such Plan

15

according to the actuarial assumptions and valuations most recently used (to the extent the assumptions comply with SFAS 87) and no transaction contemplated by this Agreement shall cause the sum of each such Plan's assets, insured obligations or reserves to be less than such Plan's benefit obligations. As used in this Agreement, the term "Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlled by, or under common control with such Person.

- (x) Simultaneously with or prior to the execution of this Agreement, the Board of Directors of the Company and the Compensation and Stock Option Committee of the Board have adopted resolutions approving the 2006 Executive Incentive Plan, which plan is set forth on Schedule 4.1(i)(x) of the Company Disclosure Schedules (the "Executive Incentive Plan").
- (xi) Notwithstanding any other provision of this Agreement to the contrary, except as ERISA and any other Laws applicable to the Company Compensation and Benefits Plans are implicated by Sections 4.1(d), (e), (f), (g) or (h), the representations and warranties contained in this Section 4.1(i) constitute the sole representations and warranties of the Company with respect to ERISA and any other Laws applicable to the Company Compensation and Benefits Plans.
- (j) Compliance with Laws; Permits. Except as disclosed in the Company Reports or as set forth on Schedule 4.1(j) of the Company Disclosure Schedules, neither the Company nor any of its Subsidiaries has violated or failed to comply with any Law, and to the knowledge of the Company, neither the Company nor any of its Subsidiaries is under investigation with respect to or been threatened to be charged with or given notice of any violation of any Law, except for any such violations or failures to comply that have not had, and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. The Company and each of its Subsidiaries has all permits, licenses, authorizations, consents, approvals and franchises from Governmental Entities required to conduct its businesses as presently being conducted, except for those the absence of which has not had, and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
  - (k) Anti-takeover Statutes; Rights Agreement; Appraisal Rights.
- (i) The Company has taken all actions necessary to exempt the Merger, this Agreement and the other transactions contemplated hereby from the provisions of Sections 607.0901 and 607.0902 of the FBCA. No other "control share acquisition," "fair price," "moratorium" or other anti-takeover Laws enacted by U.S. state or federal Governmental Entities apply to this Agreement or any of the transactions contemplated hereby.
  - (ii) The Company has taken all action necessary to render the Rights Agreement inapplicable to the Merger and the transactions contemplated hereby.
  - (iii) No holder of Shares nor any other Person will have any appraisal or dissenters' rights with respect to any Shares, pursuant to the FBCA or any other provision of

- (l) Environmental Matters. (i) Except as disclosed in the Company Reports, as set forth on Schedule 4.1(l) of the Company Disclosure Schedules and for such matters that have not had, and could not reasonably be expected to have, either individually or in the aggregate, a Company Material Adverse Effect: (A) the Company and its Subsidiaries and their respective operations are in compliance with all applicable Environmental Laws; (B) the Company and its Subsidiaries possess and maintain in effect and are in compliance with all environmental permits, licenses, authorizations and approvals required under applicable Environmental Laws with respect to their properties and businesses; (C) neither the Company nor any of its Subsidiaries has received any written claim, notice or request for information concerning the violation of or liability under any applicable Environmental Law; (D) there are no writs, injunctions, decrees, orders, judgments or Litigation Claims outstanding or pending, or to the knowledge of the Company, threatened, relating to compliance by the Company or any of its Subsidiaries with any Environmental Law or any environmental permits, licenses, authorizations and approvals required under applicable Environmental Laws or to any liability of the Company or any of its Subsidiaries under any applicable Environmental Law; and (E) there are no liabilities of or relating to the Company or any Subsidiary, whether absolute, accrued, contingent, determined or determinable, arising under or relating to any Environmental Law.
- (ii) There has been no material environmental investigation, study, audit, test, review or other analysis conducted of which the Company or any Subsidiary of the Company possesses or has control in relation to the current or prior business of the Company or any Subsidiary of the Company or any property or facility now or previously owned, leased or operated by the Company or any such Subsidiary which has not been delivered or made available to Parent prior to the date hereof.
- (iii) Neither the Company nor any Subsidiary owns, leases or operates any real property in New Jersey or Connecticut or, in connection with the transactions contemplated hereby, would otherwise be subject to the filing or other requirements of the environmental transfer laws of such states.

Notwithstanding any other provision of this Agreement to the contrary, except as environmental matters are implicated by Sections 4.1(d), (e) or (g), the representations and warranties of the Company contained in this Section 4.1(l) constitute the sole representations and warranties of the Company with respect to environmental matters.

As used herein, the terms (i) "Environmental Law" shall mean any Law relating to the protection of the environment, human health and safety and/or pollutants, contaminants, wastes or chemicals or any toxic, radioactive, ignitable, corrosive, reactive or otherwise hazardous substances, wastes or materials and (ii) "Company" and "Subsidiary" shall include any entity which is, in whole or in part, a predecessor of the Company or any Subsidiary.

(m) Tax Matters. Except as set forth on Schedule 4.1(m) of the Company Disclosure Schedules:

17

- (i) The Company and its Subsidiaries (A) have duly filed (taking into account any extension of time within which to file) all material Tax Returns required to be filed by any of them, which Tax Returns are true, correct and complete in all material respects; (B) have paid all Taxes that are shown as due on such filed Tax Returns and any other material Taxes that the Company or any of its Subsidiaries are otherwise obligated to pay, except with respect to Taxes that are being contested in good faith, and no material penalties or charges are due with respect to the late filing of any Tax Return required to have been filed by or with respect to any of them, have not waived any statute of limitations with respect to Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency; (D) do not have any deficiency, audit, examination, investigation or other proceeding in respect of material Taxes or Tax matters pending or, to the Company's knowledge, threatened; and (E) have provided adequate reserves, in accordance with U.S. GAAP, as reflected in the most recent consolidated financial statements of the Company and its Subsidiaries contained in the Company Reports for any material Taxes of the Company or any of its Subsidiaries that have not been paid.
- (ii) Neither the Company nor any of its Subsidiaries is a party to, is otherwise bound by or has an obligation under, any Tax sharing agreement, Tax indemnification agreement, Tax allocation agreement or similar Contract or arrangement.
- (iii) Neither the Company nor any of its Subsidiaries shall be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (A) change in method of accounting prior to the Closing Date under Code Section 481; (B) "closing agreement" as described in Code Section 7121 (or any corresponding or similar provision of state, local or foreign income Tax Law) executed on or prior to the Closing Date; or (C) installment sale or intercompany transaction (as defined in Treasury Regulation Section 1502-13) made on or prior to the Closing Date. As used herein, the term "Treasury Regulations" shall mean the rules and regulations promulgated by the U.S. Treasury Department under the Code.
- (iv) Each of the Company and its Subsidiaries has withheld and paid to the appropriate Taxing authority all material Taxes required to have been withheld and paid by it in connection with amounts paid or owing to any current or former employee, independent contractor, creditor or other third party and has complied in all material respects with all applicable Laws relating to the payment and withholding of such Taxes.
- (v) Neither the Company nor any of its Subsidiaries has any material liability for the Taxes of any Person (other than the Company and its Subsidiaries) under Treasury Regulation Section 1.1502 -6 (or any corresponding or similar provision of state, local or foreign Tax Law), as a transferee or successor, by contract or otherwise.
- (vi) The Company has made available to Parent true and correct copies of the U.S. federal income Tax Returns filed by the Company and its Subsidiaries for the Tax years 2001 through 2004.

18

- (vii) There is no Contract, plan or arrangement of the Company or its Subsidiaries covering any Person that, individually or collectively, would constitute compensation in excess of the deduction limitation set forth in Code Section 162(m), except as disclosed in the Company Reports.
- (viii) Neither the Company nor any of its Subsidiaries has in any year for which the applicable statute of limitations remains open distributed capital stock of another Person, or has had its capital stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Code Sections 355 or 361.
- (ix) Neither the Company nor any of its Subsidiaries has entered into any agreement or arrangement with any Governmental Entity with regard to the Tax liability of the Company or any Subsidiary affecting any Tax period for which the applicable statute of limitations, after giving effect to extensions or waivers, has not expired.

- (x) As used herein, the terms (A) "Tax" (including, with correlative meaning, the term "Taxes,") shall include all federal, state, local and foreign income, profits, franchise, gross receipts, environmental, capital stock, payroll, sales, use, employment, unemployment, disability, property, transfer, documentary, stamp, registration, recording, withholding, excise, production, value added, occupancy and other taxes, fees, duties or assessments of any nature whatsoever, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of such penalties and additions, and (B) "Tax Return" shall include all returns and reports (including elections, declarations, disclosures, schedules, estimates and information returns, as well as attachments thereto and amendments thereof) required to be provided to a Governmental Entity relating to Taxes.
- (xi) None of the Tax Returns referred to in Subsection (vi) above and, to the Company's knowledge, no other Tax Return, contains any position that is, or would be, subject to penalties under Code Section 6662 (or any corresponding provisions of state, local or non-U.S. Tax Law). Neither the Company nor any of its Subsidiaries has entered into any "listed transactions" as defined in Treasury Regulation 1.6011 -4(b)(2), and the Company and its Subsidiaries have properly disclosed all reportable transactions as required by Treasury Regulation 1.6011-4, including filing Forms 8886 with Tax Returns and with the Office of Tax Shelter Analysis.
- (xii) None of the assets of the Company or any of its Subsidiaries is property that the Company or any Subsidiary is required to treat as being a "safe harbor lease" within the meaning of Code Section 168(f)(8), as in effect prior to amendment by the Tax Equity and Fiscal Responsibility Act of 1982.
- (xiii) None of the assets of the Company or any of its Subsidiaries has been financed with or directly or indirectly secures any debt the interest on which is tax-exempt under Code Section 103(a). Neither the Company nor any Subsidiary is a borrower or guarantor of any outstanding industrial revenue bonds, and neither the Company nor any Subsidiary is a tenant, principal user or related person to any principal user (within the meaning of Code Section 144(a)) of any property that has been financed or improved with the proceeds of any industrial revenue bonds.

19

- (xiy) None of the assets of the Company or any of its Subsidiaries is "tax exempt use property" within the meaning of Section 168(h) of the Code.
- (xv) Neither the Company nor any of its Subsidiaries has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897 of the Code.
  - (xvi) Neither the Company nor any of its Subsidiaries has made an election to calculate its liability for California Franchise Tax according to the "water's edge" method.

Notwithstanding any other provision of this Agreement to the contrary, except as Tax matters are implicated by Sections 4.1(d), (e), (f), (g), (h) or (j), the representations and warranties of the Company contained in this Section 4.1(m) constitute the sole representations and warranties of the Company with respect to Tax matters.

- (n) [Intentionally Deleted]
- (o) <u>Labor Matters</u>. Except as set forth on <u>Schedule 4.1(o)</u> of the Company Disclosure Schedules: (i) (A) neither the Company nor any of its Subsidiaries is the subject of any proceeding asserting that the Company or any of its Subsidiaries has committed an unfair labor practice or any other violation of Law relating to employee matters except for any that, individually or in the aggregate, are not, and could not reasonably be expected to be, material to the Company and its Subsidiaries taken as a whole, (B) since December 31, 2003, there has not been any labor strike, walk-out, work stoppage, slow-down or lockout involving employees of the Company or any of its Subsidiaries and (C) since December 31, 2003, no labor organization or group of employees has filed any representation petition or made any written or oral demand for recognition.
- (ii) The Company and its Subsidiaries have at all times since December 31, 2002 and in all material respects properly classified each of its employees as employees and each of its independent contractors as independent contractors, as applicable, and no notice has been received from any Governmental Entity that such contractors would be considered employees for employment Law or Tax purposes.
- (iii) As of the date hereof, within the last three years, neither the Company nor any of its Subsidiaries has effectuated (A) a "plant closing" (as defined in the Worker Adjustment and Retraining Notification Act (the "WARN Act")) affecting any site of employment or one or more facilities or operating units within any site of employment or facility of the Company or any of its Subsidiaries; (B) a "mass layoff" (as defined in the WARN Act or any foreign equivalent); or (C) engaged in layoffs or employment terminations or other similar terminations sufficient to trigger application of any similar state, local or foreign Law that has had or could reasonably be expected to have a Company Material Adverse Effect.
  - (p) Intellectual Property. Except as set forth on Schedule 4.1(p) of the Company Disclosure Schedules:

20

- (i) The Company and/or each of its Subsidiaries owns, or is licensed or otherwise has legally enforceable rights to use, all material Company Intellectual Property Rights, free and clear of any Lien (other than licenses granted in the ordinary course of business consistent with past practices). There exist no material restrictions (other than as imposed by applicable Law, or as otherwise imposed by the Company under applicable confidentiality agreements or by the Company with respect to the Company's trade secrets) on the disclosure, use, license or transfer of the Company Intellectual Property Rights owned by the Company or any of its Subsidiaries. The consummation of the transactions contemplated by this Agreement will not alter, encumber, impair or extinguish in any material respect any of the Company Intellectual Property Rights. As used in this Section 4.1(p), the term "Company Intellectual Property Rights" shall mean all inventions, whether or not patentable, patents, patent applications, designs, trademarks, trade names, service marks, logos, brand names, domain names (and all goodwill associated with the foregoing) copyrights, computer software, trade secreted information and know-how (whether tangible or intangible), databases and data collections and any other intellectual property rights or similar intangible rights that are currently used or held for use in the business of the Company or any of its Subsidiaries.
- (ii) Except as has not had and could not reasonably be expected to have a Company Material Adverse Effect, neither the use of the Company Intellectual Property Rights by the Company or any of its Subsidiaries nor the conduct of the business of the Company or any of its Subsidiaries conflicts with, infringes upon, violates or interferes with, or

constitutes an appropriation of, any right, including any patent, trademark, trade name, service mark or copyright or other intellectual property right, of any other Person. There is no claim, action, suit, investigation or proceeding pending against, or to the knowledge of the Company, threatened against or affecting the Company or any of its Subsidiaries (A) challenging the validity of the Company's or any of its Subsidiaries' ownership of, or right to use, any material Company Intellectual Property Rights (and the Company otherwise has no knowledge that any material Company Intellectual Property Right is invalid or conflicts with any asserted right of any other Person); (B) or seeking to deny or restrict the rights of the Company or any Subsidiary in any of the Company Intellectual Property Rights; or (C) alleging that the Company or any of its Subsidiaries have infringed, misappropriated or otherwise violated any third-party intellectual property rights. None of the Company Intellectual Property Rights material to the operation of the business of the Company or any of its Subsidiaries has been adjudged invalid or unenforceable.

- (iii) To the knowledge of the Company, there are no infringements by third parties of any material Company Intellectual Property Rights.
- (iv) Schedule 4.1(p)(iv) of the Company Disclosure Schedules contains a true and complete list of (A) all registrations or applications for registrations included in the Company Intellectual Property Rights that are owned by the Company or any of its Subsidiaries and (B) all agreements (excluding licenses for commercial off the shelf computer software that are generally available on nondiscriminatory pricing terms which have an acquisition cost of \$50,000 or less) to which the Company or any of its Subsidiaries is a party or otherwise bound and pursuant to which (x) the Company or any of its Subsidiaries obtains the right to use or a covenant not to be sued under, any intellectual property right and/or (y) the Company or any of

21

its Subsidiaries grants the right to use, or a covenant not to be sued under, any intellectual property right.

- (q) Title to Properties. Except as disclosed in the Company Reports, the Company and its Subsidiaries have good (and in the case of real property, fee simple, or the local equivalent) title to, or a valid leasehold interest in, their material properties and assets, except for defects or imperfections of title, easements, restrictive covenants and similar encumbrances that, in the aggregate, do not materially interfere with the ability of the Company and its Subsidiaries, taken as a whole, to conduct their businesses as presently conducted. Except as disclosed in the Company Reports, all such material properties and assets are held free and clear of all mortgages, liens, pledges, charges, security interests, encumbrances or other adverse claims of any kind in respect of such properties or assets (collectively, "Liens"), except for (i) Liens for Taxes not yet due and payable or being contested in good faith and for which adequate accruals or reserves have been established on the consolidated balance sheet of the Company as of September 30, 2005 in the Company Reports; (ii) Liens of materialmen, mechanics, carriers, landlords and like Persons that individually or in the aggregate do not materially interfere with any present use of such properties or assets by the Company or its Subsidiaries; (iii) in the case of any leased properties or assets, Liens that affect the landlord's interest under the applicable lease and that individually or in the aggregate do not materially interfere with any present use of such properties or assets by the Company or its Subsidiaries; (iv) Liens disclosed in the title insurance policies or surveys made available to the Parent with respect to owned real property and that individually or in the aggregate do not materially interfere with any present use of such properties or assets by the Company or its Subsidiaries; and (v) such Liens that, individually or in the aggregate, have not had, and could not reasonably be expected to have, a Company Material Adverse Effect. The Company has delivered or made available to the Parent, with respect to each of the leased real properties, copies of the applicable leases and all material written extensions, amendments, supplements and other modifications thereof and has provided accurate summaries of all material oral extensions, amendments, supplements and other modifications. All leases pursuant to which the Company or its Subsidiaries leases from other Persons material items of real or personal property are valid and enforceable against the Company or its Subsidiaries in accordance with their respective terms, subject to the Bankruptcy and Equity Exception; and there is no default or event of default under any of such leases of the Company or any of its Subsidiaries or, to the knowledge of the Company, any other party thereto (or any condition that with notice or lapse of time, or both, would constitute a default), that has had, or could reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (r) <u>Material Contracts</u>. Except as set forth on <u>Schedule 4.1(r)</u> of the Company Disclosure Schedules or as disclosed in the Company Reports (including all exhibits thereto (including exhibits incorporated by reference therein)), neither the Company nor any of its Subsidiaries is a party to or otherwise bound by:
  - (i) any lease of personal property providing for annual rental payments of \$1,000,000 or more or any lease of any material real property;
  - (ii) any agreement (other than purchase orders entered into in the ordinary course of business) for the purchase of materials, supplies, goods, services, equipment

22

or other assets (including specifically-manufactured or unique parts) the terms of which provide for aggregate payments by the Company and its Subsidiaries of \$1,000,000 or more;

- (iii) any sales, distribution or other similar agreement (other than purchase orders entered into in the ordinary course of business) providing for the sale by the Company or any of its Subsidiaries of materials, supplies, goods, services, equipment or other assets that provides for aggregate payments to the Company and its Subsidiaries of \$1,000,000 or more:
  - (iv) any partnership, joint venture or other similar agreement or arrangement;
  - (v) any agreement relating to the acquisition or disposition of any business (whether by merger, sale of stock, sale of assets or otherwise);
- (vi) any agreement relating to indebtedness for borrowed money or the deferred purchase price of property (in either case, whether incurred, assumed, guaranteed or secured by any asset);
  - (vii) any license, franchise or similar agreement that is not terminable by the Company without material penalty upon ninety (90) days or less notice to the other party thereto;
- (viii) any agreement (or provision in an agreement) whose purpose is to limit the freedom of the Company or any of its Subsidiaries to compete in any line of business or with any Person or in any area and which limitation is material or which would so limit the freedom of the Company or any of its Subsidiaries after the Closing Date;

- (ix) any agreement with (A) any Person directly or indirectly owning, controlling or holding with power to vote, 5% or more of the outstanding voting securities of the Company or any of its Subsidiaries, (B) any Person (other than the Company and its Subsidiaries) 5% or more of whose outstanding voting securities are directly or indirectly owned, controlled or held with power to vote by the Company or any of its Subsidiaries or (C) any director or officer of the Company or any of its Subsidiaries or any "associates" or members of the "immediate family" (as such terms are respectively defined in Rule 12b-2 and Rule 16a-1 of the Exchange Act) of any such director or officer;
- (x) any material agreement to indemnify a third party other than any agreement entered into in the ordinary course of business consistent with past practices or any other agreement listed in any sub-section of Schedule 4.1(r) of the Company Disclosure Schedules; or
  - (xi) any other Contract not made in the ordinary course of business that is material to the Company and its Subsidiaries, taken as a whole.

Each agreement, contract, plan, lease, arrangement or commitment disclosed or required to be disclosed on <u>Schedule 4.1(r)</u> of the Company Disclosure Schedules (each, a "<u>Material Contract</u>") is a valid and binding agreement of the Company or a Subsidiary of the Company, as

23

the case may be, and is in full force and effect. True and complete copies of each Material Contract (including any material modifications and amendments thereto and material waivers thereunder) have been made available to Parent. Neither the Company nor any of its Subsidiaries, nor, to the knowledge of the Company, any other party thereto, is in material violation of, or in material default under, nor does there exist any condition that with notice or lapse of time, or both, would cause such a material violation of or material default under, any Material Contract to which it is a party or by which it, or any of its properties or assets, is otherwise bound.

- (s) <u>Insurance</u>. The Company maintains insurance policies covering its (and its Subsidiaries') assets, business, properties, operations, employees, directors and officers, and product warranty and liability claims, and such other forms of insurance in such amounts, with such deductibles and against such risks and losses as are, in its judgment, reasonable for the business and assets of the Company and its Subsidiaries, taken as a whole. Except as set forth on <u>Schedule 4.1(s)</u> of the Company Disclosure Schedules, all such insurance policies are in full force and effect and the Company is otherwise in compliance with such policies and such policies will not, by their terms, terminate or lapse by reason of this Agreement or the consummation of the transactions contemplated hereby, except for any failures to so be in force and effect or comply or terminations or lapses that, individually or in the aggregate, have not had, and could not reasonably be expected to have, a Company Material Adverse Effect.
- (t) <u>Vote Required</u>. The approval of the holders of a majority of the outstanding Shares as of the applicable record date (the "<u>Company Requisite Vote</u>") is the only vote of any class or series of the capital stock of the Company required to approve this Agreement, the Merger and the other transactions contemplated hereby.
- (u) <u>Brokers and Finders</u>. Except for Lehman Brothers Inc., neither the Company nor, to the Company's knowledge, any of its officers, directors or employees has engaged any financial advisor, broker or finder or incurred or will incur any liability for any financial advisory fees, brokerage fees, commissions or finders' fees in connection with this Agreement, the Merger and the other transactions contemplated hereby.
- (v) Opinion of Financial Advisor. The Board of Directors of the Company has received the opinion of Lehman Brothers Inc., on or prior to the date of this Agreement, to the effect that, as of the date of such opinion and based upon and subject to the matters set forth therein, the Merger Consideration to be received by the holders of Shares is fair, from a financial point of view, to such holders, a signed copy of which opinion has been delivered to Parent.
- (w) Information to be Supplied; Disclosure Documents. The proxy statement of the Company to be filed with the SEC in connection with the Merger (the "Company Proxy Statement") and any amendments or supplements thereto will, when filed, comply in all material respects with the applicable requirements of the Exchange Act and the rules and regulations thereunder and all other applicable Laws. At the time the Company Proxy Statement or any amendment or supplement thereto is first mailed to stockholders of the Company, and at the time such stockholders vote on adoption of this Agreement, the Company Proxy Statement, as supplemented or amended, if applicable, will not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in the

24

light of the circumstances under which they were made, not misleading. The representations and warranties contained in this Section 4.1(w) will not apply to statements or omissions included in the Company Proxy Statement based upon information furnished to the Company in writing by Parent specifically for use therein.

- (x) Foreign Corrupt Practices Act and International Trade Sanctions. Neither the Company nor any of its Subsidiaries, nor any of their respective directors, officers, agents or employees in connection with the business of or services performed for the Company or its Subsidiaries: (i) has promised, offered, given or authorized the giving of anything of value to a any officer or employee of any government or government-controlled entity or of a public international organization, or any person acting in an official capacity for or on behalf of any of the foregoing, or any political party or official thereof, or candidate for political office (a "Government Official"), or to any other person while knowing that all or some portion of the money or value will be offered, given or promised to a Government Official for the purposes of securing any improper or undue advantage; (ii) has accepted or received any unlawful contributions, payments, expenditures or gifts; (iii) is the subject of, or does or in the past three years has done business in or with a country or persons that is the subject of, sanctions imposed or administered by the U.S. Treasury Department's Office of Foreign Assets Control, the United Nations Security Council, or the Governmental Entities of a jurisdiction in which the Company or a Company Subsidiary is organized or doing business (collectively, "Economic Sanctions"), or has otherwise violated or operated in noncompliance with any applicable export restrictions, anti-boycott regulations or other applicable domestic or foreign Laws, except, in each case, as failure to comply is not and could not reasonably be expected to be, individually or in the aggregate, material to the Company and its Subsidiaries, taken as a whole.
- (y) No Other Representations or Warranties. Except for the representations and warranties contained in this Section 4.1, neither the Company nor any other Person makes any other express or implied representation or warranty on behalf of or in respect of the Company or any of its Subsidiaries.
  - 4.2 Representations and Warranties of Parent and Merger Sub. Parent and Merger Sub hereby jointly and severally represent and warrant to the Company that:

(a) Organization, Good Standing. Each of Parent and Merger Sub is an entity duly organized, validly existing and in good standing (with respect to jurisdictions that recognize
such concept) under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own and operate its material properties
and assets and to carry on, in all material respects, its business as presently conducted.

(b) <u>Corporate Authority</u>. Each of Parent and Merger Sub has the requisite corporate or similar power and authority, and has taken all action necessary in order, to execute, deliver and perform its obligations under this Agreement and to consummate, upon the terms and subject to the conditions of this Agreement, the transactions contemplated hereby, except for obtaining the Parent Required Statutory Approvals. This Agreement has been duly executed and delivered by Parent and Merger Sub and, assuming due authorization, execution and delivery by the Company, constitutes the valid and binding obligation of each of Parent and Merger Sub,

25

enforceable against each of them in accordance with its terms, subject to the Bankruptcy and Equity Exception.

## (c) Governmental Filings, No Violations.

- (i) Except for any reports, filings, registrations, approvals and notices required to be made: (A) pursuant to Section 1.2, (B) under the HSR Act and the Exchange Act, (C) as set forth on Schedule 4.2(c)(i) of the Parent Disclosure Schedules (items (B) and (C), the "Parent Required Statutory Approvals"), and (D) by Parent or Merger Sub on behalf of itself and the Company in Italy with the Italian Competition Authority, no notices, reports, registrations or other filings are required to be made by Parent or Merger Sub with, nor are any consents, approvals, permits or authorizations required to be obtained by Parent or Merger Sub from, any U.S. or foreign Governmental Entity in connection with the execution and delivery by Parent and Merger Sub of this Agreement and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby, other than those notices, reports, registrations or filings that the failure to make has not had, and could not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- (ii) The execution, delivery and performance of this Agreement by Parent and Merger Sub do not, and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby will not, constitute or result in (A) a breach or violation of the articles of incorporation or by-laws (or comparable governing instruments) of Parent or Merger Sub or (B) a breach or violation of, or constitute a default under, any contract or agreement to which Parent or Merger Sub is a party or that it is otherwise bound by or any Law to which Parent or Merger Sub is subject, except, in the case of clause (B) above, for any breaches, violations or defaults that have not had, and could not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.

As used herein, the term "Parent Material Adverse Effect" shall mean a material adverse effect on the ability of Parent and/or Merger Sub to execute and deliver or perform its obligations under this Agreement or to consummate the transactions contemplated hereby.

- (d) <u>Litigation</u>. There are no Litigation Claims pending or, to the knowledge of Parent, threatened against Parent, Merger Sub or any of its other Subsidiaries that, if adversely determined or resolved, could reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- (e) Compliance with Laws. Neither Parent nor any of its Subsidiaries has violated or failed to comply with any Laws, except where any such violations or failures to comply have not had, and could not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- (f) No Business Activities. Merger Sub is not a party to any material agreements nor has it conducted any activities other than in connection with its organization, the negotiation and execution of this Agreement and the consummation of the transactions contemplated hereby.

26

- (g) <u>Brokers and Finders</u>. Except for Goldman, Sachs & Co., the fees, commissions and expenses of which will be paid by Parent, neither Parent, Merger Sub nor, to Parent's knowledge, any of their respective officers, directors or employees has engaged any broker or finder or incurred any liability for any brokerage fees, commissions or finders' fees in connection with the Merger, this Agreement and the other transactions contemplated hereby.
- (h) <u>Financial Capability.</u> Parent has the financial and other capabilities to perform, and to cause Merger Sub and the Surviving Corporation to perform, its and their obligations in accordance with this Agreement. Parent has available cash or cash equivalents sufficient to enable it to fund the Merger Consideration when and as set forth in Article III and any other amounts payable by Parent, Merger Sub and the Surviving Corporation contemplated by this Agreement.
- (i) <u>Board and Stockholder Approval</u>. The Boards of Directors of Parent and Merger Sub have duly approved the Merger and this Agreement. No action is necessary under applicable Law or Parent's organizational instruments on the part of the stockholders of Parent in connection with this Agreement or the Merger. Parent, as the sole stockholder of Merger Sub, has approved the Merger and this Agreement.
- (j) Information to be Supplied. None of the information to be supplied by Parent or Merger Sub in writing specifically for inclusion or incorporation by reference in the Company Proxy Statement or any amendment or supplement thereto, at the time the Company Proxy Statement or any amendment or supplement thereto is first mailed to stockholders of the Company and at the time the stockholders vote on adoption of this Agreement, will contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.
- (k) No Other Representations or Warranties. Except for the representations and warranties contained in this Section 4.2, neither Parent, Merger Sub nor any other Person makes any other express or implied representation or warranty on behalf of or in respect of Parent or Merger Sub.

ithheld, conditioned or delayed:  (a) the Company and its Subsidiaries	shall conduct their respective bu	sinesses in the ordinary course a	nd in a manner consistent with past pra	actice:
(a) the company and no sucondantes	January Control Control	27	a in a manner consistent with past pra	
		_,		

- (b) the Company shall not: (i) amend its articles of incorporation or by-laws or the comparable governing instruments of any of its Subsidiaries; (ii) split, combine, subdivide, or reclassify its outstanding shares of Company Common Stock; (iii) declare, set aside or pay any dividend of cash, stock or property in respect of the Company Common Stock; or (iv) repurchase, redeem or otherwise acquire any shares of its capital stock or any securities convertible into or exchangeable or exercisable for any shares of its capital stock or permit any of its Subsidiaries to do so (other than as may be required pursuant to the Indenture or the Convertible Notes);
- (c) except as set forth on Schedule 5.1(c) of the Company Disclosure Schedules, neither the Company nor any of its Subsidiaries shall issue, sell, pledge or otherwise encumber any shares of its capital stock, or any securities convertible into or exchangeable or exercisable for, shares of its capital stock or any Voting Debt (other than shares of Company Common Stock issuable pursuant to the (i) exercise of Company Options outstanding as of December 31, 2005 (whether or not presently vested); (ii) conversion of the Convertible Notes; or (iii) exercise of the Finestar Warrant);
- (d) neither the Company nor any of its Subsidiaries shall, other than in the ordinary course of business or in connection with any of the restructuring initiatives disclosed in the Company Reports (including the sale of the manufacturing facility in Hungary contemplated thereby), lease, license, sell, mortgage or dispose of any property or assets or amend, supplement or otherwise modify in any material respect or extend or terminate any material lease or license of real property;
- (e) neither the Company nor any of its Subsidiaries shall effect any material acquisition of the assets, business, capital stock or other equity of any Person (whether by way of merger, consolidation, tender offer, share exchange or other activity), except for purchases of inventory, equipment and raw materials in the ordinary course of business;
- (f) neither the Company nor any of its Subsidiaries shall cancel or forgive any material indebtedness owed to the Company or any of its Subsidiaries (other than indebtedness owing by the Company to any of its wholly-owned Subsidiaries or vice versa);
- (g) neither the Company nor any of its Subsidiaries shall adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, recapitalization or other similar reorganization;
- (h) except as may be required to comply with applicable Law, neither the Company nor any of its Subsidiaries shall terminate, adopt or increase the benefits provided under any Company Compensation and Benefit Plans, except for: (i) increases occurring in the ordinary course of business consistent with past practice (which shall include normal periodic performance reviews and related compensation and benefit increases); and (ii) actions necessary to satisfy existing obligations under Company Compensation and Benefit Plans, Contracts and other legal obligations existing as of the date hereof;

28

- (i) except as may be required by applicable Law or to the extent required by U.S. GAAP (as advised by the Company's independent accountants), neither the Company nor any of its Subsidiaries shall change any material accounting principle, practice or method;
  - (j) neither the Company nor any of its Subsidiaries shall incur, create, assume or guarantee any material indebtedness for borrowed money;
- (k) except as set forth on Schedule 5.1(k) of the Company Disclosure Schedules, neither the Company nor any of its Subsidiaries shall settle or compromise any material Tax liability, make or amend any Tax election, or amend any material Tax Return;
- (l) neither the Company nor any of its Subsidiaries shall make any loans, advances or capital contributions to, or investments in, any other Person other than (i) by the Company or any of its wholly-owned Subsidiaries or (ii) as required by any Contract or other legal obligation existing as of the date hereof;
- (m) neither the Company nor any of its Subsidiaries shall settle or compromise any action, suit, claim, litigation, proceeding, arbitration, investigation, audit or controversy ("Proceeding") or enter into any consent, decree, injunction or similar restraint or form of equitable relief in settlement of a Proceeding, except for such settlements, compromises, consents, decrees, injunctions, or similar restraints or forms of equitable relief that are not material to the Company and its Subsidiaries, taken as a whole; and
  - (n) neither the Company nor any of its Subsidiaries will authorize or enter into an agreement to do anything prohibited by the foregoing subsections.
- 5.2 Additional Covenants of the Company. Subject to Section 6.1, prior to the Company or any of its Subsidiaries (i) amending in any material respect or terminating any Material Contract or (ii) entering into any Contract that if entered into prior to the date hereof would constitute a Material Contract, the Company shall notify Parent of its intention to take any of the foregoing actions and shall reasonably consult with Parent in connection therewith.

#### ARTICLE VI

## ADDITIONAL AGREEMENTS

6.1 Access. The Company agrees that, throughout the period from the date hereof until the Effective Time or the earlier termination of this Agreement, except as may otherwise be required or restricted by applicable Law, it shall (and shall cause its Subsidiaries to) (a) afford Parent's officers, employees, counsel, accountants and other authorized

representatives reasonable access, during normal business hours and upon reasonable prior notice to its officers, employees, properties, books, Contracts and records, (b) furnish promptly to Parent such financial and operating and other information with respect to its business and properties as may reasonably be requested, but only to the extent that such access does not unreasonably interfere with the business or operations of the Company or its Subsidiaries and (c) instruct its Representatives to cooperate with Parent in its investigation; provided, however, that the Company and its Subsidiaries shall not be required to provide information (including pursuant to

29

Section 5.2) in breach of applicable Law or that is subject to existing confidentiality obligations. All requests for information made pursuant to this Section 6.1 shall be directed to an executive officer of the Company or its financial advisor or such other Person as may be designated by the Company's executive officers. Notwithstanding the foregoing, Parent shall not have access to the Company's or its Subsidiaries' proprietary or competitively sensitive information (including pursuant to Section 5.2) if the Company reasonably believes that such information cannot be made available to Parent in a manner that does not cause the Company or its Subsidiaries to be materially and competitively disadvantaged. All such information shall be governed by the terms of the Confidentiality Agreement. No information or knowledge obtained in any investigation pursuant to this Section 6.1 shall affect or be deemed to modify any representation or warranty made by the Company hereunder.

6.2 Acquisition Proposals. (a) The Company shall not, and shall not permit or authorize any of its Subsidiaries or any of its or their officers or directors to, and shall cause the officers, directors or employees, agents and representatives (including accountants, attorneys and investment bankers) of the Company and its Subsidiaries (collectively, the "Representatives") not to, directly or indirectly: (A) initiate, solicit, encourage or otherwise facilitate any inquiries or the making of any proposal or offer with respect to (i) a merger, consolidation, business combination, share exchange, sale of all or substantially all of the assets, reorganization, recapitalization, liquidation, dissolution or other similar transaction involving the Company or any of its Subsidiaries, (ii) the acquisition of a significant equity interest (20% or more) in the Company or any of its Subsidiaries, whether by tender offer (including a self-tender offer), exchange offer or otherwise or (iii) the acquisition of a significant portion (20% or more on a consolidated basis) of the assets of the Company and its Subsidiaries (any such proposal or offer described in clauses (i) - (iii) (other than the transactions contemplated by this Agreement) being hereinafter referred to as an "Acquisition Proposal"); (B) engage in any negotiations concerning, or furnish any information or data to, or have any discussions (other than a statement that refers only to this Section and the Company's agreement not to engage in further discussions) with, or otherwise cooperate, assist or participate in any effort by any Person relating to an Acquisition Proposal; (C) approve, recommend or endorse an Acquisition Proposal, or fail to make the Company Board Recommendation, or withdraw or modify in a manner adverse to Parent the Company Board Recommendation (any of the foregoing in this clause (C), an "Adverse Recommendation Change"); (D) enter into any letter of intent or similar agreement with respect to any class of equity securities of the Company or any

(b) Nothing contained in Section 6.2(a) or elsewhere in the Agreement shall prevent the Company or its Representatives from: (A) furnishing information or data to, or entering into discussions or negotiations with, a third party in response to a *bona fide* written Acquisition Proposal made by such third party (the initial submission of which to the Company was not, directly or indirectly, initiated, solicited, encouraged or otherwise facilitated by the Company, its Subsidiaries, or any of their respective Representatives), provided that prior to the Company's furnishing any confidential information to such party, such party shall have entered into a confidentiality agreement with terms no less favorable to the Company than those contained in the Confidentiality Agreement; (B) following receipt of a *bona fide* written

30

Acquisition Proposal made by another party (the initial submission of which to the Company was not, directly or indirectly, initiated, solicited, encouraged or otherwise facilitated by the Company, its Subsidiaries, or any of their respective Representatives), (i) making an Adverse Recommendation Change as a result of a Superior Proposal or (ii) immediately prior to a termination of this Agreement by the Company pursuant to Section 8.1(f), entering into an agreement, letter of intent or similar document relating to a Superior Proposal; (C) complying with Rule 14e-2 (and any associated obligations under Rule 14d-9) of the Exchange Act with respect to a third-party tender or exchange offer (provided that the Board of Directors of the Company shall not recommend acceptance of such tender or exchange offer unless the conditions to making an Adverse Recommendation Change pursuant to clause (B) have been satisfied); (D) taking any action ordered to be taken pursuant to a non-appealable final order by any court of competent jurisdiction; or (E) making any disclosure or filing, in its reasonable judgment after receiving advice from outside counsel (which may be its regularly engaged counsel), that is required by Law (including the FBCA and the rules and regulations, orders or requests of any Governmental Entity (including the SEC)); provided, however, that in each case referred to in the foregoing clauses (A) and (B) only if (1) the Acquisition Proposal is a Superior Proposal at the time that the Company determines to take such action(s) or, in the case of an action pursuant to clause (A), the Board of Directors of the Company shall have concluded in good faith that such Acquisition Proposal is reasonably likely to result in a Superior Proposal; (2) the Board of Directors of the Company shall determine in good faith, after consultation with outside legal counsel (which may be its regularly engaged counsel), that such action(s) is necessary in order for the Board of Directors to comply with its fiduciary duties to the Company's stockholders under applicable Law; and (3) the Company Requisite Vote has not yet been obtained; provided, further, that the Board of Directors of the Company shall not take any of the actions referred to in the foregoing clause (B) until at least three Business Days shall have elapsed following Parent's receipt of written notice from the Company advising Parent that the Company's Board of Directors has received a Superior Proposal and (i) intends to make an Adverse Recommendation Change as a result of a Superior Proposal or (ii) intends to terminate this Agreement pursuant to Section 8.1(f) and enter into an agreement, letter of intent or similar document with respect to such Superior Proposal, attaching the most current version of any proposed agreement or a reasonably detailed summary of the material terms and conditions of such Superior Proposal and the identity of the offeror (it being understood and agreed that any material amendment or modification of such Proposal shall result in a new three-Business Day period), and Parent does not make, during such three-Business Day period(s), an irrevocable offer that the Company's Board of Directors shall have concluded, in its good faith judgment, after consultation with its financial advisor and outside legal counsel, is at least as favorable (taking into account, without limitation, the proposed transaction structures) to the Company's stockholders as such Superior Proposal. The Company agrees that during such three-Business Day period, the Company and its Representatives shall negotiate in good faith with Parent and its Representatives regarding any revisions proposed by Parent to the terms of the transactions contemplated by this Agreement.

The Company will promptly notify Parent, in writing, of the existence of any proposal, discussion, negotiation or inquiry received by the Company with respect to any Acquisition Proposal, and the Company will promptly communicate to Parent the material terms and conditions of any such proposal, discussion, negotiation or inquiry that it may receive, whether orally or in writing, and the identity of the Person making such proposal or inquiry or

engaging in such discussion or negotiation, which information shall be subject to the Confidentiality Agreement. The Company will promptly provide to Parent any non-public information concerning the Company provided to any other Person that shall not have been previously provided to Parent, which information shall be subject to the Confidentiality Agreement. The Company will keep Parent reasonably informed of the status of any such Acquisition Proposal (including modifications or proposed modifications thereto), which information shall be subject to the Confidentiality Agreement.

Without prejudice to any actions permitted to be taken by the Company pursuant to this Section 6.2(b), the Company shall, and shall cause its Representatives to, immediately cease and cause to be terminated any existing discussions or negotiations with any Persons conducted heretofore with respect to any Acquisition Proposal and shall, in accordance with the Company's rights and subject to the Company's obligations under applicable confidentiality agreements, use its reasonable efforts to cause any such Person (or its agents or advisors) in possession of confidential information of the Company that was furnished by or on behalf of the Company in connection with a potential sale of the Company to return or destroy all such information (including any such information incorporated into any analysis prepared by any such Persons or any of their advisors) and shall terminate any access such Person may have been given by the Company or its Representatives to any "virtual dataroom" of the Company. The Company agrees that it will take the necessary steps to promptly inform any Persons referred to in the first sentence of this paragraph of the obligations of the Company in this Section 6.2.

As used herein, the term "Superior Proposal" shall mean an unsolicited bona fide written Acquisition Proposal to acquire, directly or indirectly, more than fifty percent (50%) of the Company Common Stock then outstanding or a majority of the assets of the Company and its Subsidiaries, taken as a whole, in a single transaction or a series of related transactions, and otherwise on terms that the Company's Board of Directors determines in good faith (after consultation with its financial advisor and outside legal counsel which may be its regularly engaged counsel), (i) to be more favorable from a financial point of view to the Company's stockholders than the Merger (or any subsequent offer made by Parent in response to such Acquisition Proposal) and (ii) is fully financed or for which financing is committed or which, in the good faith judgment of the Company's Board of Directors is capable of being obtained by such Person.

6.3 Stockholders Meeting. The Company shall cause a meeting of its stockholders (the "Company Stockholders Meeting") to be duly called and held as soon as reasonably practicable for the purpose of voting on the approval and adoption of this Agreement and the Merger. Subject to Section 6.2(b), the Board of Directors of the Company shall recommend approval and adoption of this Agreement and the Merger by the Company's stockholders. In connection with such meeting, the Company shall (i) promptly prepare and file with the SEC, use its reasonable best efforts to have cleared by the SEC and thereafter mail to its stockholders as promptly as practicable the Company Proxy Statement and all other proxy materials for such meeting, (ii) use its reasonable best efforts to obtain the Company Stockholder Approval and (iii) otherwise comply with all legal requirements applicable to such meeting. Parent and its counsel shall be afforded a reasonable opportunity to review and comment upon the Company Proxy Statement prior to its being filed with the SEC or being disseminated to the holders of Shares. The Company shall provide Parent and its counsel, in writing, with any

32

comments that the Company or its counsel may receive from the SEC or its staff with respect to the Company Proxy Statement promptly after their receipt of such comments.

- 6.4 Filings; Other Actions; Notification. (a) Each party hereto shall file or cause to be filed as promptly as practicable after the date hereof with the Federal Trade Commission and the Department of Justice and other applicable Governmental Entities all notifications required to be filed under the HSR Act and any other applicable Laws, in accordance with the rules and regulations under such applicable Laws, with respect to the transactions contemplated hereby. Each party hereto will use reasonable best efforts to make such filings promptly and to respond on a timely basis to any requests for additional information made by any such Governmental Entity.
- (b) The Company and Parent (and Merger Sub) shall fully cooperate with each other and use (and shall cause their respective Representatives and Subsidiaries to use) reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, necessary or advisable on its part under this Agreement and applicable Laws to consummate the Merger and the other transactions contemplated hereby as soon as practicable, including preparing and filing as soon as practicable all documentation to effect all necessary notices, applications, reports and other filings and to obtain, as soon as practicable, all Company Required Statutory Approvals and Parent Required Statutory Approvals, as the case may be, and all consents, approvals and authorizations necessary or advisable to be obtained from any Person in order to consummate the Merger or any of the other transactions contemplated hereby, including with or from any works counsel, labor union or similar entity or governing body. Parent (and Merger Sub) and the Company shall fully cooperate with each other in connection with the making of all such notices, applications, reports and filings, including providing copies of all such documents to the non-filing party and its advisors prior to filing, and, if requested, to accept all reasonable additions, deletions or changes suggested in connection therewith; provided that the parties hereto understand and agree that the reasonable best efforts of any party hereto shall not be deemed to include (i) entering into any settlement, undertaking, consent decree, stipulation or agreement with any Governmental Entity in connection with the transactions contemplated hereby or (ii) divesting or otherwise holding separate (including by establishing a trust or otherwise) (or otherwise agreeing to do any of the foregoing) with respect to Parent or the Company or any of their Subsidiaries or any of their respective businesses, assets or properties. Subject to applicable Laws relating to the exchange of information an
- (c) Subject to any confidentiality obligations and the preservation of any attorney-client or similar privilege, the Company and Parent (and Merger Sub) shall use reasonable best efforts to keep the other apprised of the status of matters relating to completion of the transactions contemplated hereby, including promptly furnishing the other with copies of notices or other communications received by Parent (and Merger Sub) or the Company, as the case may be, from any Person and/or Governmental Entity with respect to the Merger and the other transactions contemplated hereby.

and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order, whether temporary, preliminary or permanent, that shall be in effect and that prohibits, prevents or materially restricts consummation of any of the transactions contemplated hereby.

- 6.5 <u>Vote of Shares Owned by Parent</u>. Parent shall cause all Shares owned by Parent, Merger Sub and any other Subsidiary of Parent to be duly voted in favor of the Merger and this Agreement at the Company Stockholders Meeting.
- 6.6 <u>Publicity.</u> The initial press release regarding the transactions contemplated hereby shall be a joint press release and, thereafter, the Company and Parent shall consult with the other party prior to issuing any press releases or otherwise making public announcements with respect to the Merger and the other transactions contemplated hereby and prior to making any filings with any Person and/or Governmental Entity with respect thereto, except as may otherwise be required by Law or by obligations pursuant to any listing agreement with or the rules of any national securities exchange or national market quotation system.
- 6.7 <u>Employee Benefits Matters; Bonus Plan.</u> (a) Except as otherwise provided in this Section 6.7, the Surviving Corporation shall provide, during the period commencing at the Effective Time and ending on the first anniversary thereof, to the current (as of the Effective Time) employees of the Company and its Subsidiaries ("<u>Affected Employees</u>") (i) at least the same rate of base pay, commissions and bonus opportunity as those currently provided under the Company Compensation and Benefit Plans and (ii) other benefits (other than equity-based benefits or arrangements) that, in the aggregate, are equivalent, or superior, to those currently provided under the Company Compensation and Benefit Plans.
- (b) At and following the Effective Time, Parent shall credit all service by Affected Employees of the Company and its Subsidiaries (and any predecessor entities) prior to the Effective Time to be taken into account for all purposes (other than for benefit accrual purposes under any defined benefit pension plan) under any compensation and benefit plans (including any equity-based plans), programs, practices and policies of Parent or its Subsidiaries (the "Benefit Plans") in which the Affected Employees shall participate.
- (c) From and after the Effective Time, Parent shall, with respect to all Affected Employees participating in the Benefit Plans, use reasonable best efforts to (i) cause, to the extent such conditions would not apply or would have been waived under the Company Compensation and Benefit Plans, to be waived any pre-existing condition limitations and any waiting period limitations under employee welfare benefit plans, policies or practices of Parent or its Subsidiaries in which employees of the Company or its Subsidiaries participate and (ii) cause to be credited any deductibles, co-payment amounts and out-of-pocket expenses incurred

34

by such employees and their beneficiaries and dependents during the portion of the calendar year prior to participation in the Benefit Plans provided by Parent and its Subsidiaries.

- (d) Parent shall cause the Surviving Corporation to honor all employee benefit obligations as of the Effective Time to current and former employees and directors of the Company under the Company Compensation and Benefit Plans and shall cause the Surviving Corporation to pay all fees and expenses due to directors and all retirement benefits due or to become due under the Artesyn Technologies, Inc. Outside Directors Retirement Plan and agreements pursuant thereto in accordance with the terms thereof. Parent shall cause the Surviving Corporation to assume the obligations under the employment and severance agreements and change of control agreements to which the Company is a party or otherwise bound by as of the Effective Time, including the agreements set forth on Schedule 6.7(d) of the Company Disclosure Schedules.
- 6.8 Indemnification; Directors' and Officers' Insurance; Exculpation. (a) The Surviving Corporation shall indemnify and hold harmless (x) to the fullest extent permitted under applicable Law and (y) without limiting clause (x), as required pursuant to any existing indemnification agreements or obligations of the Company (and the Surviving Corporation also shall advance attorneys' fees and expenses as incurred to the fullest extent permitted under applicable Law and as required pursuant to any existing indemnification agreements or obligations of the Company; provided, that the Person to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined that such Person is not entitled to indemnification), each present and former director and officer of the Company and its Subsidiaries (collectively, the "Indemnified Parties") against any costs or expenses (including attorneys' fees and expenses), judgments, fines, losses, claims, settlements, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to matters existing or occurring at or prior to the Effective Time and relating to their service to and activities for the Company and/or its Subsidiaries, including the transactions contemplated hereby.
- (b) As of the Effective Time, the Surviving Corporation shall have purchased, and fully prepaid, all premiums in respect of, and shall thereafter maintain in effect, directors' and officers' liability insurance coverage for the Company's present and former directors and officers for a period of six years after the Effective Time that provides for runoff coverage in an amount at least equal to that presently provided by the Company for its directors and officers and otherwise on terms that are the same as, in all material respects, those presently provided by the Company (as disclosed to Parent prior to the date hereof); provided, that, in satisfying its obligations under this Section 6.8(b), the Surviving Corporation shall not be obligated to pay an aggregate premium in excess of 250% of the amount the Company paid with respect to its last full fiscal year, which amount the Company has disclosed to Parent prior to the date hereof; provided, further, that the Company may, prior to the Effective Time and after reasonably consulting with Parent, purchase and fully prepay such premiums in an aggregate amount that shall not exceed 250% of the amount the Company paid with respect to its last full fiscal year.
- (c) The By-Laws of the Surviving Corporation shall include provisions for indemnification on the same basis as set forth in the Company's By-laws in effect on the date hereof. For six years after the Effective Time, Parent shall cause the Surviving Corporation to

35

maintain in effect the provisions in its By-laws providing for indemnification and exculpation from liability of Indemnified Parties with respect to circumstances and events occurring or existing at or prior to the Effective Time, to the fullest extent permitted from time to time under the FBCA, which provisions shall not be amended, except only as (and only to the extent) required by applicable Law.

(d) The rights of each Indemnified Party under this Section 6.8 shall be in addition to any rights that such Person may have under the articles of incorporation or by-laws of the Company or any of its Subsidiaries, applicable Law (including the FBCA) or any agreement of an Indemnified Party with the Company or any of its Subsidiaries. Such rights shall be cumulative and not exclusive. The provisions of this Section 6.8 are intended to be for the benefit of, and shall be enforceable by, each of the Indemnified Parties, their respective heirs, legal beneficiaries and representatives.

- (e) In the event that the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or other entity of such consolidation or merger or (ii) transfers or conveys all or substantially all of its properties or assets to another Person, then, and in each such case, to the extent necessary to effect the purposes of this Section 6.8, proper provision shall be made so that the successors and assigns of the Surviving Corporation assume the obligations set forth in this Section 6.8. The provisions of this Section 6.8 shall survive the Effective Time.
- 6.9 Executive Incentive Plan. (a) Promptly after the date hereof, the Company shall communicate and explain the terms of the 2006 Executive Incentive Plan to all employees covered by such Plan and shall use reasonable best efforts to cause all such employees to be bound by the terms thereof.
  - (b) The Company agrees that, without Parent's consent, it shall not make and shall not agree to make any changes to the terms of the Executive Incentive Plan.
- 6.10 <u>Transfer Tax</u>. Parent and the Company shall cooperate in the preparation, execution and filing of all returns, applications, questionnaires or other documents, regarding any real property transfer, stamp, recording, documentary, sales, use, value added, stock transfer and any other Taxes that shall become payable in connection the Merger (collectively, "<u>Transfer Taxes</u>"). From and after the Effective Time, the Surviving Corporation shall pay or cause to be paid all Transfer Taxes.

#### ARTICLE VII

### CONDITIONS

- 7.1 Conditions to Each Party's Obligation to Effect the Merger. The respective obligation of each party to effect the Merger is subject to the satisfaction at or prior to the Effective Time of each of the following conditions:
  - (a) Stockholder Approvals. The Company Requisite Vote shall have been obtained.

36

- (b) HSR Act. The waiting period under the HSR Act applicable to the consummation of the Merger shall have expired or been earlier terminated.
- (c) Other Approvals. (i) All Company Required Statutory Approvals and Parent Required Statutory Approvals shall have been obtained at or prior to the Effective Time and (ii) all other approvals, the failure of which to obtain could reasonably be expected to have a Parent Material Adverse Effect and/or a Company Material Adverse Effect, shall have been obtained at or prior to the Effective Time.
- (d) <u>Injunction</u>. No order, stay, decree, judgment or injunction shall have been entered, issued or enforced by any Governmental Entity or court of competent jurisdiction and remain in effect that prohibits consummation of the Merger, and there shall not have been any action taken, or any Law enacted, entered, enforced or deemed applicable to the Merger that makes the consummation of the Merger illegal or prohibits the Merger.
- 7.2 <u>Conditions to Parent's and Merger Sub's Obligation to Effect the Merger</u>. The obligations of Parent and Merger Sub to effect the Merger are subject to the satisfaction (or written waiver) of each of the following additional conditions:
- (a) <u>Bringdown of Representations and Warranties; Performance of Obligations</u>. (i) The Company shall have performed in all material respects all of its obligations hereunder required to be performed by it at or prior to the Effective Time, (ii) the representations and warranties of the Company contained in Sections 4.1(b) and (c) shall be true in all material respects at and as of the Effective Time as if made at and as of such time (except to the extent such representations and warranties speak as of an earlier time, in which case such representations and warranties shall be true in all material respects at and as of such earlier time), (iii) the other representations and warranties of the Company contained in this Agreement, disregarding all qualifications and exceptions contained therein relating to materiality or Company Material Adverse Effect, shall be true at and as of the Effective Time as if made at and as of such time (except to the extent such representations and warranties speak as of an earlier time, in which case such representations and warranties shall be true at and as of such earlier time), with only such failures to so be true as have not had and could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect and (iv) Parent shall have received a certificate signed by an executive officer of the Company to the foregoing effect.
- (b) No Suits or Challenges. There shall not have been instituted or pending by any Person or Governmental Entity any suit, action or proceeding that is reasonably likely to prevail, (i) challenging or seeking to make illegal, restrain, prohibit or delay materially the consummation of the Merger, (ii) seeking to limit the ownership or operation by the Company or Parent of a material portion of the business, operations or assets of the Company and its Subsidiaries, taken as a whole (or an amount of Parent and its Subsidiaries that would be material to the Company and its Subsidiaries, taken as a whole) or (iii) seeking to compel Parent or any of its Subsidiaries to dispose of or hold separate all or any material portion of the business or assets of the Company and its Subsidiaries, taken as a whole);

37

provided, however, that prior to (and as a condition to) invoking this condition, Parent shall have complied with Section 6.4.

- (c) No Governmental Restrictions. No statute, rule, regulation, judgment, order or injunction shall be enacted, entered, enforced, promulgated or deemed applicable to the Merger, or any other action shall be taken by any Governmental Entity or court, other than the application to the Merger of applicable waiting periods under the HSR Act, that is reasonably likely to result in any of the consequences referred to in clauses (i) through (iii) of Section 7.2(b) above; provided, however, that prior to (and as a condition to) invoking this condition, Parent shall have complied with Section 6.4.
- (d) No Company Material Adverse Effect. No event, occurrence, revelation, development or state of circumstances or facts that, individually or in the aggregate, has had or could reasonably be expected to have a Company Material Adverse Effect shall have occurred and be continuing as of the Effective Time.

- 7.3 Conditions to the Company's Obligation to Effect the Merger. The obligation of the Company to effect the Merger is subject to the satisfaction (or written waiver) of the following additional condition:
- (a) <u>Bringdown of Representations and Warranties; Performance of Obligations</u>. (i) Each of Parent and Merger Subsidiary shall have performed in all material respects all of its obligations hereunder required to be performed by it at or prior to the Effective Time, (ii) the representations and warranties of Parent and Merger Sub contained in this Agreement, disregarding all qualifications and exceptions contained therein relating to materiality or Parent Material Adverse Effect, shall be true at and as of the Effective Time as if made at and as of such time (except to the extent such representations and warranties speak as of an earlier time, in which case such representations and warranties shall be true at and as of such earlier time) with only such failures to so be true as have not had and could not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect and (iii) the Company shall have received a certificate signed by an executive officer of Parent to the foregoing effect.

#### ARTICLE VIII

#### **TERMINATION**

- 8.1 <u>Termination</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, notwithstanding any requisite approval and adoption of this Agreement and the transactions contemplated hereby by the stockholders of the Company:
  - (a) by mutual written consent duly authorized by the Boards of Directors of each of Parent and the Company; or
  - (b) by either Parent or the Company, if:

38

- (i) the Merger shall not have been consummated on or before the date that is six months after the date of this Agreement (the "Termination Date"); provided, however, that the right to terminate under this Section 8.1(b)(i) shall not be available to any party whose failure to fulfill any of its obligations, including, without limitation, its obligations under Section 6.4, under this Agreement has been the cause of, or resulted in, the failure of the consummation of the Merger to occur on or before the Termination Date; provided, further, that, if the regulatory process necessary for a decision by the applicable Governmental Entity with respect to any antitrust clearance included in the Company Required Statutory Approvals or Parent Required Statutory Approvals has not been completed, such Termination Date may, at the option of the Company exercised in writing, be extended for a period of up to 30 days; or
- (ii) there shall be any Law that makes consummation of the Merger illegal or otherwise prohibited or if any court of competent jurisdiction or other Governmental Entity shall have issued an order, judgment, decree, ruling, injunction or taken any other action restraining, enjoining or otherwise prohibiting or materially altering the terms of the Merger and such order, judgment, decree, ruling, injunction or other action shall have become final and non-appealable, except that no party may terminate pursuant to this Section 8.1(b)(ii) unless such party shall have complied with its obligations under Section 6.4; or
  - (iii) the Company's Stockholders Meeting (including any adjournment thereof) shall have concluded and the Company Requisite Vote shall not have been obtained; or
- (c) by Parent, if (i) an Adverse Recommendation Change has been made (it being understood, however, that, for all purposes of this Agreement, the fact that the Company has provided any Person with information regarding the Company or has entered into discussions or negotiations with such Person as permitted by this Agreement shall not be deemed an Adverse Recommendation Change) or (ii) the Company shall have violated or breached, in any material respect, its obligations under Section 6.2 or Section 6.3; or
- (d) by the Company, in the event of a breach by Parent or Merger Sub of any representation, warranty or agreement contained herein where such breach would give rise to the failure of the condition set forth in Section 7.3(a) and such condition is incapable of being satisfied by the Termination Date; or
- (e) by Parent, in the event of a breach by the Company of any representation, warranty or agreement contained herein where such breach would give rise to the failure of the condition set forth in Section 7.2(a) and such condition is incapable of being satisfied by the Termination Date; or
- (f) by the Company, if (i) the Board of Directors of the Company has authorized the Company to enter into a written agreement concerning a Superior Proposal and (ii) the Company shall have complied with Section 6.2 (including Section 6.2(b) and each proviso therein applicable to clause (B) of Section 6.2(b)) and paid any amounts due pursuant to Section 8.3(a) or Section 8.3(b) in accordance with the terms, and at the times, specified therein.
  - 8.2 Effects of Termination. Except as provided in the Confidentiality Agreement and provisions of this Agreement relating thereto, in the event of the termination of

39

this Agreement pursuant to, and in accordance with, Section 8.1, this Agreement shall forthwith become void, there shall be no liability under this Agreement on the part of Parent, Merger Sub or the Company or any of their respective officers or directors and all rights and obligations of the parties hereto shall cease, subject to the sole and exclusive remedies of Parent and Merger Sub and the Company set forth in Section 8.3; provided, however, that nothing herein shall relieve any party from liability for any willful breach hereof; provided, further, that, if all conditions set forth in Article VII have been satisfied, the failure of any party hereto to effect the Merger, without terminating this Agreement in accordance herewith, shall be deemed to be a willful breach hereof. The payment of the Expenses and, if and as due, the Termination Fee by the Company or a termination fee by Parent, in each case pursuant to Section 8.3, upon a termination of this Agreement pursuant to Section 8.1 shall constitute liquidated damages in consideration of the time and opportunity and related costs of Parent and Merger Sub or the Company, as applicable, and, upon receipt of such amount(s), if and as due, except in the case of a willful breach hereof, Parent and Merger Sub shall have no further recourse at law or in equity against Parent, in respect of this Agreement or any breach hereof.

(a) The Company shall pay to Parent its Expenses upon the termination of this Agreement pursuant to (i) Section 8.1(b)(iii), if prior to the vote of the Company's stockholders on the Merger an Adverse Recommendation Change has occurred or the Company has received an Acquisition Proposal or an Acquisition Proposal has been made public, (ii) Section 8.1(c) or (iii) Section 8.1(f). As used herein, the term "Expenses" shall mean all reasonable and documented out-of-pocket expenses (excluding any wages or salaries of any of Parent's or Merger Sub's or any of their affiliates' employees) incurred by Parent and/or Merger Sub prior to the termination of this Agreement in connection with the Merger and the other transactions contemplated hereby; provided, however, that Expenses shall not exceed \$2,500,000. If the Company is required to pay Parent its Expenses upon termination of this Agreement, such Expenses shall be payable immediately on or prior, and as a condition, to the termination of this Agreement in the event of a termination by the Company and not later than two Business Days after the date of the termination of this Agreement by Parent.

(b) The Company shall pay Parent a termination fee of \$15,000,000 (the "Termination Fee") upon the termination of this Agreement pursuant to (i) Section 8.1(b)(iii) if an Adverse Recommendation Change shall have occurred prior to the vote of the Company's stockholders on the Merger, (ii) Section 8.1(c) or (iii) Section 8.1(f). If (x) this Agreement is terminated pursuant to Section 8.1(b)(iii) and prior to the vote of the Company's stockholders on the Merger no Adverse Recommendation Change shall have occurred but the Company shall have received an Acquisition Proposal or an Acquisition Proposal shall have been made public and (y) within 12 months after any such termination the Company enters into, or announces its intent to enter into, an Acquisition Proposal with a third party, the Company shall pay Parent the Termination Fee upon the Company's entry into, or announcement of an intent to enter into, such Acquisition Proposal. If (A) this Agreement is terminated pursuant to Section 8.1(b)(i) and prior to such termination, the Company shall have received an Acquisition Proposal or an Acquisition Proposal shall have been made public after the receipt of the Company Requisite Vote and (B) within 12 months after any such termination the Company enters into, or announces its intent to

40

enter into, such Acquisition Proposal, the Company shall pay Parent the Termination Fee upon the consummation of such Acquisition Proposal; provided that the Termination Fee shall not be payable pursuant to this sentence if this Agreement is terminated pursuant to Section 8.1(b)(i) and at such time any of the antitrust clearances included in the Company Required Statutory Approvals or Parent Required Statutory Approvals have not been obtained or the conditions set forth in Article VII have not been satisfied due to the failure to obtain such antitrust clearances. If the Company is required to pay Parent a Termination Fee upon termination of this Agreement, such Termination Fee shall be payable immediately on or prior, and as a condition, to the termination of this Agreement by Parent. If the Company is required to pay Parent a Termination Fee pursuant to the second and third sentences of this Section 8.3(b), such fee shall be payable immediately on or prior, and as a condition, to the last event triggering such obligation.

- (c) If this Agreement is terminated pursuant to Section 8.1(b)(i) and the waiting period under the HSR Act applicable to the consummation of the Merger shall not have expired or been earlier terminated on or prior to such date of termination or any of the antitrust clearances included in the Company Required Statutory Approvals or Parent Required Statutory Approvals have not been obtained on or prior to such date of termination, but all other conditions set forth in Article VII have been satisfied (except to the extent not satisfied due to the failure to obtain antitrust clearances), then Parent shall pay the Company a termination fee of \$15,000,000. If Parent is required to pay the Company a termination of this Agreement, such termination fee shall be payable immediately on or prior, and as a condition, to the termination of this Agreement in the event of a termination by Parent and not later than two Business Days after the date of the termination of this Agreement by the Company.
- (d) Any Termination Fee and Expenses required to be paid pursuant to this Section 8.3 shall be paid by wire transfer of immediately available funds to an account designated by Parent or as designated by the Company if a fee is paid pursuant to Section 8.3(c).
- (e) Except as otherwise provided in Sections 6.10 and 8.2 and this Section 8.3, all costs and expenses incurred in connection with this Agreement and the Merger and any other transactions contemplated hereby shall be paid by the party incurring such costs and expenses, whether or not the Merger or any other transaction contemplated hereby is consummated.

## ARTICLE IX

## MISCELLANEOUS AND GENERAL

9.1 Non-Survival of Representations, Warranties and Agreements. None of the representations, warranties or covenants (subject to the immediately succeeding sentence) in this Agreement or in any document or instrument delivered pursuant to this Agreement shall survive the Effective Time. This Section 9.1 shall not limit any covenant or agreement of the parties that by its terms contemplates or permits performance after the Effective Time.

41

- 9.2 Modification or Amendment. Subject to the provisions of applicable Law, at any time prior to the Effective Time, the parties hereto may modify or amend this Agreement, by action authorized by their respective boards of directors and a written agreement executed and delivered by duly authorized executive officers of the respective parties.
- 9.3 Waiver of Conditions. The conditions to each of the parties' obligations to consummate the Merger contained herein are for the sole benefit of such party and may be waived only by such party, in whole or in part, but only if and to the extent permitted herein and by applicable Law.
- 9.4 <u>Counterparts</u>. This Agreement may be executed in counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement. This Agreement may be executed by facsimile signature, which for all purposes hereunder shall have the same force and effect as an original.
- 9.5 Governing Law and Venue. THIS AGREEMENT SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAW OF THE STATE OF NEW YORK APPLICABLE TO AGREEMENTS ENTERED INTO AND TO BE WHOLLY PERFORMED IN SUCH STATE, EXCEPT TO THE EXTENT THAT MATTERS PROVIDED FOR HEREIN ARE REQUIRED TO BE GOVERNED BY THE FBCA OR U.S. FEDERAL LAW. The parties hereby irrevocably submit to the exclusive jurisdiction of the courts of the State of New York (except if a matter is subject to the exclusive jurisdiction of the U.S. federal courts) and the U.S. federal courts located, in each case, in the Borough of Manhattan in respect of the interpretation and enforcement of this Agreement, the documents referred to in this Agreement and the transactions contemplated hereby and thereby. The parties hereby waive, and agree not to assert, as a defense in any action, suit or proceeding in respect of the interpretation or enforcement hereof or of any such document that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in said courts or that the venue thereof

is not appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims with respect to such action, suit or proceeding shall be heard and determined in such State of New York or U.S. federal courts. The parties hereby consent to and grant, to the fullest extent permitted by applicable Law, any such court jurisdiction over the person of such parties and over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 9.7 or in such other manner as may be permitted by Law shall be valid and sufficient service thereof

9.6 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

9.7 Notices. Any notice, request, instruction or other document required or permitted to be given hereunder by any party hereto to the others shall be in writing and delivered personally, sent by registered or certified mail or reputable overnight courier (e.g., Federal Express, DHL), postage prepaid, or transmitted by facsimile (upon receipt of electronic or telephonic confirmation of successful transmission):

42

if to Parent or Merger Sub, Emerson Electric Co.

8000 W. Florissant Avenue P.O. Box 4100

St. Louis, MO 63136-8506 Attention: Frank J. Dellaquila Facsimile: (314) 553-1605

and

Michael Keating, Esq. Facsimile: (314) 553-1232

with a copy to (which shall not constitute notice):

Davis Polk & Wardwell 450 Lexington Avenue New York, NY 10017

Attention: Christopher Mayer, Esq.

Facsimile: (212) 450-3800

if to the Company,

Artesyn Technologies, Inc.
7900 Glades Road
Suite 500
Boca Raton, Florida 33434
Attention: Chief Executive Officer

Facsimile: (561) 451-1020

with a copy to (which shall not constitute notice):

Kirkpatrick & Lockhart Nicholson Graham LLP 599 Lexington Avenue New York, NY 10022

Attention: John D. Vaughan, Esq. and Whitney J. Smith, Esq.

Facsimile: (212) 536-3901

or to such other Persons or addresses as may be designated in writing by the party to receive such notice as provided earlier in this Section 9.7. Any such notice shall be effective (i) upon receipt if personally delivered or if delivered postage prepaid, return receipt requested, or if sent by a nationally recognized overnight courier service, or (ii) on the date of transmission if such notice is sent by facsimile prior to 5:00 p.m. (New York time) on a Business Day, or otherwise on the Business Day after the date of transmission.

9.8 Entire Agreement; No Other Representations. This Agreement (including the Company Disclosure Schedules and the Parent Disclosure Schedules) and the Confidentiality

REPRESENTATIONS OR WARRANTIES MADE BY ITSELF, OR ANY OF ITS RESPECTIVE OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, FINANCIAL AND LEGAL ADVISORS OR OTHER REPRESENTATIVES, WITH RESPECT TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY, NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE OTHER OR THE OTHER'S REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION WITH RESPECT TO ANY OF THE FOREGOING.

- 9.9 <u>Binding Effect; No Third-Party Beneficiaries</u>. This Agreement shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and permitted assignees. Other than with respect to the matters set forth in Section 6.8 (Indemnification; Directors' and Officers' Insurance; Exculpation) and in Article III, this Agreement is not intended to confer upon any Person other than the parties hereto (and their respective successors and permitted assignees) any rights or remedies hereunder.
- 9.10 Obligations of Parent and the Company. Except as otherwise specifically provided herein, whenever this Agreement requires Merger Sub to take any action, such requirement shall be deemed to include an undertaking on the part of Parent to cause Merger Sub to take such action and, after the Effective Time, to cause the Surviving Corporation to take such action. Whenever this Agreement requires a Subsidiary of the Company to take any action, such requirement shall be deemed to include an undertaking on the part of the Company to cause such Subsidiary to take such action.
- 9.11 Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of any other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall, subject to Section 9.5, such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.
- 9.12 <u>Disclosure Schedules</u>. Any information disclosed in any section of the Company Disclosure Schedules or the Parent Disclosure Schedules shall be deemed to be disclosed in any other section(s) of the Company Disclosure Schedules or the Parent Disclosure Schedules, respectively, where it is reasonably apparent that the disclosure contained in such section of the Company Disclosure Schedules or the Parent Disclosure Schedules, as applicable, is relevant to such other section(s). The Company Disclosure Schedules and the Parent

44

Disclosure Schedules, respectively, and the information and disclosure contained therein, are intended only to qualify and clarify the representations, warranties and covenants of the Company or Parent (and Merger Sub), as the case may be, contained in this Agreement and shall not be deemed to expand in any way the scope, construction or effect of any of such representations, warranties or covenants. No reference to or disclosure of any item or other matter in the Company Disclosure Schedules or the Parent Disclosure Schedules shall be construed to establish a standard of materiality or knowledge.

- 9.13 Interpretation. The table of contents and headings herein are for convenience of reference only, do not constitute parts of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to a Section, Article or Schedule, such reference shall be to a Section or Article of or Schedule to this Agreement unless otherwise clearly indicated. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." "Dollars" or "\$" shall mean the currency of the United States that, as at the time of payment, is legal tender for the payment of public and private debts.
- 9.14 <u>Assignment</u>. This Agreement and the rights, interests or obligations hereunder shall not be assignable or delegable without the prior written consent of the other parties; <u>provided, however</u>, that Parent may designate, by written notice to the Company, another wholly owned subsidiary to be a constituent corporation in lieu of Merger Sub, so long as such designation could not reasonably be expected to: (i) impose any material delay in the obtaining of, or significantly increase the risk of not obtaining, any Parent Required Statutory Approval or Company Required Statutory Approval or termination of any applicable waiting period; (ii) significantly increase the risk of any Governmental Entity entering an order prohibiting the consummation of the Merger; (iii) significantly increase the risk of not being able to remove any such order on appeal or otherwise; or (iv) materially delay the consummation of the Merger. If the requirements of the previous sentence are met and Parent desires to designate another wholly owned subsidiary to be the constituent corporation in lieu of Merger Sub, then all references herein to Merger Sub shall be deemed references to such other subsidiary, except that all representations and warranties made herein with respect to Such other subsidiary as of the date of such designation.

[SIGNATURE PAGE TO FOLLOW]

45

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by duly authorized executive officers of the parties hereto as of the date first written above.

ARTESYN TECHNOLOGIES, INC.

By: /s/ Joseph M.O'Donnell

Name: Joseph M. O'Donnell Title: Chief Executive Officer

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Senior Vice President -

Acquisitions and Development

ATLANTA ACQUISITION SUB, INC.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Vice President

#### VOTING AGREEMENT

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and JANA Partners LLC ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. *Voting Agreement*. Stockholder hereby agrees to vote or cause to be voted or exercise or cause to be exercised its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement, and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or, subject to Section 5.03, any amended version thereof) or the Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that it will not vote or cause to be voted any Shares in favor of, nor consent to nor cause consent to, and will vote such Shares against or cause such Shares to be voted against and not consent to nor cause consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. Subject to the next sentence, the proxy granted by Stockholder pursuant to this Article 1 is

1

irrevocable during the term of this Agreement and is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be revoked upon termination of this Agreement in accordance with its terms.

# ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. Authorization. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement and the consummation by Stockholder of the transactions contemplated hereby are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement and the consummation of the transactions contemplated hereby do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree, (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder or (iv) result in the imposition of any Lien on any asset of Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner of and has the power to direct the voting of the Shares, free and clear of any Lien and any other limitation or restriction (including any restriction on the right to vote or otherwise dispose of the Shares). None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares, the options to acquire Shares, the warrants and the convertible notes set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital

2

stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company or (iii) options or other rights to acquire from the Company any capital stock, voting securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

Section 2.05. Finder's Fees. Except as provided in Sections 4.1(u) and 4.2(g) of the Merger Agreement, no investment banker, broker, finder or other intermediary is entitled to a fee or commission from Parent or the Company in respect of this Agreement based upon any arrangement or agreement made by or on behalf of Stockholder.

Parent represents and warrants to Stockholder:

Section 3.01. Corporate Authorization. The execution, delivery and performance by Parent of this Agreement and the consummation by Parent of the transactions contemplated hereby are within the corporate powers of Parent and have been duly authorized by all necessary corporate action. This Agreement constitutes a valid and binding agreement of Parent

# ARTICLE 4 COVENANTS OF STOCKHOLDER

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Except pursuant to the terms of this Agreement, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares during the term of this Agreement. Stockholder shall not seek or solicit any such sale, assignment, transfer, encumbrance or other disposition or any such contract, option or other arrangement or understanding and agrees to notify Parent promptly, and to provide all details requested by Parent, if Stockholder shall be approached or solicited, directly or indirectly, by any Person with respect to any of the foregoing.

3

Section 4.02. Other Offers. Stockholder shall not, directly or indirectly, take any action that the Company or its Representatives are prohibited from taking pursuant to Section 6.2 of the Merger Agreement. Stockholder will promptly advise and update Parent after receipt by Stockholder of an Acquisition Proposal in accordance with the notice provisions applicable to the Company as set forth in Section 6.2 of the Merger Agreement.

# ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement; provided that Stockholder shall not be required to incur any material expenses in furtherance of the foregoing.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. *Termination*. This Agreement shall terminate upon the earliest of (a) the date of termination of the Merger Agreement, (b) the Board's making of an Adverse Recommendation Change in compliance with Section 6.2 of the Merger Agreement and (c) the date that the Merger Agreement is amended to reduce the merger consideration without Stockholder's consent or otherwise amended in a manner that materially and adversely affects the Stockholder without Stockholder's consent, and all rights and obligations of the parties under this Agreement shall immediately terminate, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to such termination.

Section 5.04. Expenses. All costs and expenses incurred in connection with this Agreement shall be paid by the party incurring such cost or expense.

Section 5.05. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

4

Section 5.06. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York.

Section 5.07. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.08. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.09. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.10. Capitalized Terms. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement.

Section 5.11. *Notices*. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.12. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the record holder and

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

### EMERSON ELECTRIC CO.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila
Title: Senior Vice President –
Acquisitions and Development

## JANA PARTNERS LLC

By: /s/ Barry Rosenstein

Name: Barry Rosenstein Title: Managing Partner

### Address for Notices:

JANA Partners LLC 200 Park Avenue, Suite 3300 New York, New York 10166 Attention: General Counsel Facsimile: 212-692-7695

## Appendix A

Class of Security	Number Owned
Common Stock	4,361,065
Company Options	0
Finestar Warrants	0
Convertible Notes	0

#### VOTING AGREEMENT

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Edward S. Croft III ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

# ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

# ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Section 3.01. Corporate Authorization. The execution, delivery and performance by Parent of this Agreement are within the corporate powers of Parent and have been duly authorized by all necessary corporate action. This Agreement constitutes a valid and binding agreement of Parent.

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila
Title: Senior Vice President –

Acquisitions and Development

EDWARD S. CROFT III

By: /s/ Edward S. Croft III

Name: Edward S. Croft III

Title:

Address for Notices:

Edward S. Croft III 500 Argonne Drive, N.W. Atlanta, GA 30305

6

Class of Security	Number Owned
Common Stock	14,481
Company Options	80,000

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Gary R. Larsen ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Senior Vice President -Acquisitions and Development

GARY R. LARSEN

By: /s/ Gary R. Larsen

Name: Gary R. Larsen Title:

Address for Notices:

Gary R. Larsen 21376 Crest Falls Court Boca Raton, FL 33428

6

	Class of Security	Number Owned
Common Stock		17,110
Company Options		27,500

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Lawrence J. Matthews ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc. ("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

## ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02 . *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02 . Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03 . Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04 . *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company or (iii) options or other rights to acquire from the Company any capital stock, voting

Parent represents and warrants to Stockholder:

Section 3.01. Corporate Authorization. The execution, delivery and performance by Parent of this Agreement are within the corporate powers of Parent and have been duly authorized by all necessary corporate action. This Agreement constitutes a valid and binding agreement of Parent.

### ARTICLE 4 COVENANTS OF STOCKHOLDER

Stockholder hereby covenants and agrees that:

Section 4.01 . No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

## ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. *Governing Law.* This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10 . Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

2

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his

capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made	or
taken by Stockholder in his capacity as an officer or director of the Company.	

5

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

### EMERSON ELECTRIC CO.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Senior Vice President –

Acquisitions and Development

#### LAWRENCE J. MATTHEWS

By: /s/ Lawrence J. Matthews

Name: Lawrence J. Matthews

Title:

Address for Notices:

Lawrence J. Matthews 7601 5<sup>th</sup> Avenue South Richfield, MN 55423

6

Class of Security	Number Owned
Common Stock	743,253
Company Options	80,000

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Joseph M. O'Donnell ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc. ("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

# ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. *Non-Contravention*. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company or (iii) options or other rights to acquire from the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

### EMERSON ELECTRIC CO.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Senior Vice President –

Acquisitions and Development

JOSEPH M. O'DONNELL

By: /s/ Joseph M. O'Donnell

Name: Joseph M. O'Donnell

Title:

Address for Notices:

Joseph M. O'Donnell 3681 Carlton Place Boca Raton, FL 33496

6

<u>Appendix A</u>		
Class of Security	Number Owned	
Common Stock	303,552	
Company Options	716,500*	

<sup>\*</sup> Includes an additional 37,500 Company Options due to vest within 60 days of February 1, 2006.

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Stephen A. Ollendorff ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. *Voting Agreement*. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Senior Vice President -

Acquisitions and Development

STEPHEN A. OLLENDORFF

By: /s/ Stephen A. Ollendorff

Name: Stephen A. Ollendorff Title:

Address for Notices:

Stephen A. Ollendorff 14 Leslie Place Tenafly, NJ 07670

6

Cla	ss of Security	Number Owned
Common Stock		22,100
Company Options		90,000

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Phillip A. O' Reilly ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Senior Vice President -Title:

Acquisitions and Development

PHILLIP A. O' REILLY

By: /s/ Phillip A. O'Reilly

Name: Phillip A. O'Reilly Title:

Address for Notices:

Phillip A. O'Reilly 181 Ocean Beach Trail Vero Beach, FL 32961

6

Class of Security	Number Owned
Common Stock	111,721
Company Options	90,000

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Bert Sager ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

### ARTICLE 1 GRANT OF PROXY: VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

## ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Senior Vice President –

Acquisitions and Development

BERT SAGER

By: /s/ Bert Sager

Name: Bert Sager

Title:

Address for Notices:

Bert Sager 9000 Schoolhouse Road Miami, FL 33156

6

	Class of Security	Number Owned
Common Stock		186,267*
Company Options		90,000

<sup>\*</sup> Includes 2,080 shares of Company Common Stock, which are owned of record by Mr. Sager's wife, with respect to which Mr. Sager disclaims beneficial ownership.

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and A. Eugene Sapp, Jr. ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc. ("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Senior Vice President -Title:

Acquisitions and Development

A. EUGENE SAPP, JR.

By: /s/ A. Eugene Sapp, Jr.

Name: A. Eugene Sapp, Jr. Title:

Address for Notices:

A. Eugene Sapp, Jr. 528 Adams Street Huntsville, AL 35801

6

Class of Security	Number Owned
Common Stock	10,500
Company Options	80,000

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Ronald D. Schmidt ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc. ("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

## ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02 . *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02 . Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03 . Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04 . *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company or (iii) options or other rights to acquire from the Company any capital stock, voting

Parent represents and warrants to Stockholder:

Section 3.01. Corporate Authorization. The execution, delivery and performance by Parent of this Agreement are within the corporate powers of Parent and have been duly authorized by all necessary corporate action. This Agreement constitutes a valid and binding agreement of Parent.

### ARTICLE 4 COVENANTS OF STOCKHOLDER

Stockholder hereby covenants and agrees that:

Section 4.01 . No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

## ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. *Governing Law.* This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10 . Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

2

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his

capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.		
5		
5		
IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be du	ly executed as of the day and year first above written.	
EMI	ERSON ELECTRIC CO.	
Ву:	/s/ Frank J. Dellaquila	
	Name: Frank J. Dellaquila Title: Senior Vice President – Acquisitions and Development	
ROI	NALD D. SCHMIDT	
Ву:	/s/ Ronald D. Schmidt	
	Name: Ronald D. Schmidt Title:	
Add	Iress for Notices:	
188	nald D. Schmidt 65 64 <sup>th</sup> Avenue ppewa Falls, WI 54729	
6		

Class of Security	Number Owned
Common Stock	379,669*
Company Options	70,000

<sup>\*</sup> Includes 21,554 shares of Company Common Stock, which are owned of record by Mr. Schmidt's wife, with respect to which Mr. Schmidt disclaims beneficial ownership.

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and Lewis Solomon ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc.("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

### ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. Non-Contravention. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. Ownership of Shares. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT

Parent represents and warrants to Stockholder:

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. Survival; Termination. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto.

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

By: /s/ Frank J. Dellaquila Name: Frank J. Dellaquila Senior Vice President -Title: Acquisitions and Development LEWIS SOLOMON By: /s/ Lewis Solomon. Name: Lewis Solomon Title: Address for Notices: Lewis Solomon 79 Cherry Lane Syosset, NY 11791 6 Appendix A S Class of Security Number Owned Common Stock 11,000 Company Options 90,000

VOTING AGREEMENT (this "Agreement"), dated as of February 1, 2006 between Emerson Electric Co., a Missouri corporation ("Parent"), and John M. Steel ("Stockholder").

WHEREAS, in order to induce Parent and Atlanta Acquisition Sub, Inc. ("Merger Sub") to enter into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), with Artesyn Technologies, Inc., a Florida corporation (the "Company"), Parent has requested Stockholder, and Stockholder has agreed, to enter into this Agreement with respect to all shares of common stock, par value \$0.01 per share, of the Company that Stockholder beneficially owns (the "Shares").

NOW, THEREFORE, the parties hereto agree as follows:

# ARTICLE 1 GRANT OF PROXY; VOTING AGREEMENT

Section 1.01. Voting Agreement. Prior to the earliest to occur of (i) the Effective Time and (ii) termination of the Merger Agreement (the "Expiration Date"), Stockholder shall vote (or cause to be voted) or exercise (or cause to be exercised) its right to consent with respect to all Shares that Stockholder is entitled to vote at the time of any vote or action by written consent to approve and adopt the Merger Agreement and the Merger at any meeting of the stockholders of the Company (and at any adjournment thereof) or pursuant to actions by written consent at which such Merger Agreement (or any amended version thereof) and Merger are submitted for the consideration and vote of the stockholders of the Company. Stockholder hereby agrees that, prior to the Expiration Date, it will not vote any Shares in favor of, or consent to, and will vote (or cause to be voted) such Shares against and not consent to, the approval of any (i) Acquisition Proposal, (ii) reorganization, recapitalization, liquidation or winding-up of the Company or any other extraordinary transaction involving the Company or (iii) corporate action the consummation of which would frustrate the purposes, or prevent or delay the consummation of the transactions contemplated by the Merger Agreement.

Section 1.02. *Proxy*. Stockholder hereby revokes any and all previous proxies granted with respect to the Shares. By entering into this Agreement, Stockholder hereby grants a proxy appointing Parent as Stockholder's attorney-in-fact and proxy, with full power of substitution, for and in Stockholder's name, to vote, express consent or dissent, or otherwise to utilize such voting power in the manner contemplated by Section 1.01 above as Parent or its proxy or substitute shall, in Parent's sole discretion, deem proper with respect to the Shares. The

1

proxy granted by Stockholder pursuant to this Article 1 is granted in consideration of Parent entering into this Agreement and the Merger Agreement and incurring certain related fees and expenses. The proxy granted by Stockholder shall be irrevocable prior to the Expiration Date and shall terminate upon the Expiration Date.

## ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder represents and warrants to Parent that:

Section 2.01. *Authorization*. Stockholder has duly executed and delivered this Agreement and the execution, delivery and performance by Stockholder of this Agreement are within the powers and legal capacity of Stockholder and have been duly authorized by all necessary action. This Agreement is a valid and binding agreement of Stockholder. If Stockholder is married and the Shares set forth on Appendix A constitute community property under applicable laws, this Agreement has been duly authorized, executed and delivered by, and constitutes the valid and binding agreement of, Stockholder's spouse. If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into and perform this Agreement.

Section 2.02. *Non-Contravention*. The execution, delivery and performance by Stockholder of this Agreement do not and will not (i) if Stockholder is not a natural person, violate the organizational documents of Stockholder, (ii) violate any applicable law, rule, regulation, judgment, injunction, order or decree or (iii) require any consent or other action by any Person under, constitute a default under, or give rise to any right of termination, cancellation or acceleration or to a loss of any benefit to which Stockholder is entitled under any provision of any agreement or other instrument binding on Stockholder.

Section 2.03. *Ownership of Shares*. Stockholder is the beneficial owner and has the power to direct the voting of the Shares, free and clear of any limitation or restriction on the right to vote the Shares. None of the Shares is subject to any voting trust or other agreement or arrangement with respect to the voting of such Shares.

Section 2.04. *Total Shares*. Except for the Shares and the options to acquire Shares set forth on Appendix A, neither the Stockholder nor any of its Affiliates or "associates" (as such term is defined under Rule 12b-2 of the Exchange Act) beneficially owns any (i) shares of capital stock or voting securities of the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or voting securities of the Company any capital stock, voting

2

securities or securities convertible into or exchangeable for capital stock or voting securities of the Company.

Section 3.01. Corporate Authorization. The execution, delivery and performance by Parent of this Agreement are within the corporate powers of Parent and have been duly authorized by all necessary corporate action. This Agreement constitutes a valid and binding agreement of Parent.

### ARTICLE 4 COVENANTS OF STOCKHOLDER

Stockholder hereby covenants and agrees that:

Section 4.01. No Proxies for or Encumbrances on Shares. Prior to the Expiration Date, except pursuant to the terms of this Agreement, at the Effective Time in accordance with the terms of the Merger Agreement, or as may be required by court order or by operation of law, Stockholder shall not, without the prior written consent of Parent, directly or indirectly, (i) grant any proxies or enter into any voting trust or other agreement or arrangement with respect to the voting of any Shares or (ii) sell, assign, transfer, encumber or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the direct or indirect sale, assignment, transfer, encumbrance or other disposition of, any Shares unless the transferee of such Shares agrees in writing to be bound by the terms hereof.

### ARTICLE 5 MISCELLANEOUS

Section 5.01. Further Assurances. Parent and Stockholder will each use its reasonable efforts to execute and deliver, or cause to be executed and delivered, all further documents and instruments and use its reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary under applicable laws and regulations, to consummate and make effective the transactions contemplated by this Agreement.

Section 5.02. Amendments. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is

3

signed, in the case of an amendment, by each party to this Agreement or in the case of a waiver, by the party against whom the waiver is to be effective.

Section 5.03. *Survival; Termination*. All representations, warranties, covenants and agreements made by the parties hereto shall survive until the Expiration Date, at which time this Agreement shall be of no further force or effect, except that each party shall remain liable with respect to breaches of this Agreement occurring prior to the Expiration Date.

Section 5.04. Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other parties hereto, except that Parent may transfer or assign its rights and obligations to any Affiliate of Parent.

Section 5.05. Governing Law. This Agreement shall be construed in accordance with and governed by the laws of the State of New York, applicable to contracts entered into and fully performable within such State.

Section 5.06. Counterparts; Effectiveness. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other party hereto.

Section 5.07. Severability. If any term, provision or covenant of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 5.08. Specific Performance. The parties hereto agree that Parent and Merger Sub would suffer irreparable damage in the event any provision of this Agreement is not performed in accordance with the terms hereof and that the parties shall be entitled to specific performance of the terms hereof in addition to any other remedy to which they are entitled at law or in equity.

Section 5.09. Capitalized Terms; Beneficial Ownership. Capitalized terms used but not defined herein shall have the respective meanings set forth in the Merger Agreement. For purposes of this Agreement, "beneficial ownership" shall be determined in accordance with Rule 13d-3 under the Securities Exchange Act of 1934, as amended.

Section 5.10. Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed

4

given if delivered personally or sent by overnight courier (providing proof of delivery) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice): (i) if to Parent, to the address for notice thereto set forth in the Merger Agreement and (ii) if to Stockholder, to the address set forth on signature pagers hereto

Section 5.11. Stockholder Capacity. Notwithstanding anything else contained herein, no person executing this Agreement who is or becomes during the term hereof a director or officer of the Company makes any agreement or understanding herein in his capacity as such director or officer. Stockholder signs solely in his capacity as the beneficial owner of Stockholder's Shares and nothing in this Agreement shall limit or affect or apply to any actions, judgments or decisions made or taken by Stockholder in his capacity as an officer or director of the Company.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

### EMERSON ELECTRIC CO.

By: /s/ Frank J. Dellaquila

Name: Frank J. Dellaquila Title: Senior Vice President –

Acquisitions and Development

JOHN M. STEEL

By: /s/ John M. Steel

Name: John M. Steel

Title:

Address for Notices:

John M. Steel 211 South Helix E-54 P.O. Box 747

Solana Beach, CA 92075

6

Class of Security	Numbe rOwned
Common Stock	348,665
Company Options	40,000